

Received 30 April 2023, accepted 9 May 2023, date of publication 22 May 2023, date of current version 6 June 2023. Digital Object Identifier 10.1109/ACCESS.2023.3278261

TOPICAL REVIEW

A Review on the Applications of PSO-Based **Algorithm in Demand Side Management: Challenges and Opportunities**

FARAH ANISHAH ZAINI^{©1}, MOHAMAD FANI SULAIMA^{©1}, (Member, IEEE), INTAN AZMIRA WAN ABDUL RAZAK¹, NUR IZYAN ZULKAFLI^{D2}, AND HAZLIE MOKHLIS¹⁰³, (Senior Member, IEEE)

¹Faculty of Electrical Engineering, Universiti Teknikal Malaysia Melaka, Durian Tunggal, Malacca 76100, Malaysia ²Faculty of Mechanical Engineering, Universiti Teknikal Malaysia Melaka, Durian Tunggal, Malacca 76100, Malaysia ³Faculty of Engineering, University of Malaya, Kuala Lumpur 50603, Malaysia

Corresponding author: Mohamad Fani Sulaima (fani@utem.edu.my)

ABSTRACT The increase in energy consumption, environmental pollution issues, and low-carbon agenda has grown the research area of demand side management (DSM). DSM programs provide feasible solutions and significantly enhance the efficiency and sustainability of electrical distribution systems. This paper classifies and discusses the broad definition of DSM based on the comprehensive literature study considering demand response and energy efficiency. The implementation of Artificial Intelligence algorithms in DSM applications has been employed in many studies to help researchers make optimal decisions and achieve predictions by analyzing the massive amount of historical data. Owing to its simplicity and consistent performance in fast convergence time, Particle Swarm Optimization (PSO) is widely used as a part of the swarm AI algorithm and has become a prominent technique in the optimization process to exploit the full benefit of the demand-side program. The variants of PSO have been developed to overcome the limitations of the original PSO and solve the high complexity and uncertainty in the DSM optimization process. The proposed PSO-based algorithm can optimize consumers' consumption curves, reducing the peak demand and hence minimizing the electricity cost when integrated with the DR programs or EE measures. The research works of the PSO algorithm in DSM have seen an increasing trend in the past decade. Therefore, this paper reviewed the application of the PSO-based algorithm in DSM fields with some constraints and discussed the challenges from the previous work. The potential for new opportunities is identified so that PSO methods can be developed for future research.

INDEX TERMS Demand side management (DSM), demand response (DR), energy efficiency (EE), metaheuristic algorithms, particle swarm optimization (PSO), swarm intelligence.

I. INTRODUCTION

The increasing crisis of global warming has made it more challenging for the world to move toward a climate-resilient, low-carbon, and sustainable future in the 21st century. Burning fossil fuels and deforestation are two outcomes of human activity that have contributed to the rise in CO_2 emissions [1]. CO₂, the dominant greenhouse gas (GHG), has become a major global concern due to climate change. Based on the

The associate editor coordinating the review of this manuscript and approving it for publication was Sotirios Goudos¹⁰.

assessment report by Intergovernmental Panel on Climate Change (IPCC), the global net anthropogenic GHG emissions in 2019 have increased by about 6.5 GtCO₂-eq compared to 2010. This increase directly came from the energy supply (34%), industry (24%), agriculture, forestry, and other land use (22%), transport (15%), and buildings (6%) sectors [2]. Scientifically, the average global temperature on Earth is closely related to the concentration of GHGs in the earth's atmosphere. Global warming caused by human activities is predicted to have increased by 1.0°C above pre-industrial levels, and if it remains to rise at the current rate, it will

This work is licensed under a Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 License.

For more information, see https://creativecommons.org/licenses/by-nc-nd/4.0/

probably hit 1.5°C between 2030 and 2052 [3]. The charts in the global climate effort, including United Nation Framework Convention on Climate Change (UNFCCC), Kyoto Protocol, and Paris Agreement have reached a target to combat climate change by limiting the global average temperature to 1.5°C above pre-industrial levels by reducing CO₂ emission substantially before 2030 [4]. However, national pledges are not enough to overcome this climate issue alone. These targets would necessitate swift, extensive, and significant systemic changes, including a variety of technologies and behavioral improvements, investments in clean energy sources, and energy efficiency increases of five times by 2050 [5]. Accordingly, reducing human-caused emissions to as close to zero as possible to reach the Net-Zero emissions target around mid-century [6]. Enhancing energy efficiency and lowering energy demand are widely recognized as the quickest, safest, and cheapest ways to combat climate change [7].

In recent years, Demand Side Management (DSM) has been one of the approaches to optimize energy efficiency in electricity distribution. DSM is the practice of planning, installation, and monitoring by the electricity utility that can impact energy use by changing the consumption patterns of consumers to achieve the necessary changes in load shape. DSM aims to flatten the load profile by encouraging consumers to reduce the demand in peak hours or transfer the demand from peak to off-peak hours and preferably follow the generation pattern [8]. The main categories for DSM activities are demand response (DR) and energy efficiency (EE), as shown in Fig. 1. The demand-side solutions provide maximum benefits and are integrated with lesser risk compared to supply-side options [8], as they are potentially improving the network load pattern [9], lowering air pollution [10], reducing emission [11] and healthcare cost [12] due to its positive environmental impact. Therefore, the sustainable energy transition required a momentous effort from various scales, such as government, industry, and academia. The bright potential of DSM in industries due to the high power consumption of individuals, which encourages and facilitates involvement in DSM activities, low cost is needed because the infrastructure for metering is already in place and industrial processes are conducted in isolated locations, resulting in minimum effort for occupant comfort [13]. In line with Industrial Revolution 4.0, DSM plays a significant role in attaining efficient energy generation and utilization, particularly for transitioning business operations toward smart factories [14].

In the research area, significant references focus on the concept of DSM. A comprehensive review of the DSM, which explained the basic idea, main subjects, and practical methods by evaluating the DSM's theoretical foundation has been overviewed in [15]. The detailed issue of DSM has been emphasized in a few studies, such as demand response [16], [17], price-based program on DR concept [18], microgrid [19], energy efficiency [20], and its co-benefits [21]. Reference [22] presented the design of



FIGURE 1. Categories for DSM activities. EE is the measure to be taken by consumers, while DR is the program introduced by the utility.

DR programs and policy in European states, while strategies to accelerate the UK consumer's participation in DR have been discussed in [23] with the implementation of technologies. Furthermore, the application of DSM integrates with renewable energy [24], storage systems [25], electric vehicles (EV) [26], and building energy management systems (BEMS) [27] has been widely adopted to promote a sustainable environment and reduce environmental pollution.

The growing innovation of Internet-of-Things (IoT) monitoring, advanced metering infrastructure (AMI), automation technologies, and distributed energy resources have encouraged entities of the energy demand side to be active prosumers (producers and consumers) in the operation of the electrical grid rather than passive energy consumers. Moreover, these data acquisition technologies associated with the advancement of artificial intelligence (AI) can potentially widen the opportunities for researchers to combine interdisciplinary knowledge, examine energy consumers' operational environment in further detail, and formulate new datadriven demand-side management strategies. The appliance level energy characteristic (ALEC) can be utilized to monitor the usage of electricity appliances and energy consumption behavior closely, thus assisting the utilities and stakeholders in implementing practical DSM activities for the residential or other sectors through intrusive load monitoring (ILM) or non-ILM (NILM) techniques as discussed in [28]. Further, AI approaches forecast power demand and generation and provide better stability and efficiency for power systems [29]. On the other hand, evidence shows that developments of AI could enhance the knowledge of climate change and the modelling of its potential effects, particularly when it comes to achieving Sustainable Development Goal (SDG) number 13 on climate action. In addition, AI will also promote low-carbon energy systems with the integration of energy efficiency and renewable energy necessary to combat climate

change [30]. For instance, the various industrial sector in China could obtain the optimal path for energy efficiency enhancement which could reduce CO_2 emission by 58.31% through the prediction of the AI algorithm model [31].

Metaheuristics algorithms (MA) are a prevalent part of AI for solving optimization problems. These techniques have an acceptable performance to solve any optimization problems by finding a near-optimal solution with a limited computation burden [32]. Moreover, they are more efficient and converging than the classical approaches because of their efficiency in exploring the search space to reach a global optimum solution [33]. MA can be applied in problems with a large number of decision variables and easily adopted to a problem that has several constraints [34], such as real-world engineering design problems [35], [36]. MA optimization techniques are currently used to overcome the limitations of mathematical optimization with some great features such as fine-tuning to improve their performance, fast computing time, efficient exploration and exploitation due to nature-inspired approaches, and the independent nature of the objective function. As a result, they are effective methods for resolving optimization challenges [37], [38]. The newly formed MA methods have been very beneficial to the engineering field. Many issues, including multi-objective optimization issues as well as continuous, discrete, constrained, and other challenges, have been resolved using them. As a result of their extensive use, new scientific subfields have arisen [39]. Some of the most prominent MA found in the literature are Particle Swarm Optimization (PSO), Ant Colony Optimization (ACO), Genetic Algorithm (GA) and Differential Evolution (DE).

Numerous models have been developed with MA in the DSM optimization field. For instance, ACO based method has been proposed to optimize the residential load profile with DSM techniques while analyzing the impact of DR tariff on the user's electricity bill [40], [41]. The DE with toroidal correction has been utilized to achieve a good convergence of the algorithm and reduce the possibility of stagnation in local optima in solving the industrial load optimization considering the energy and labour costs [42]. The GA is adopted in [43] to minimize the cost and maximize the load factor simultaneously for DR residential. The newly introduced MA, namely the virulence optimization algorithm (VOA) and earthworm optimization algorithm (EWOA) are utilized individually in solving the microgrid scheduling optimization problem, which effectively shifts the energy consumption from peak to off-peak hours. As a result, the proposed method is able to minimize the mismatch between total generation and demand and the overall electricity bill [44]. Meanwhile, Grey Wolf Algorithm (GWA) shows high efficiency in solving a complex problem in energy management strategies considering hydrogen storage systems and multiple renewable sources to reduce its final operating cost [45]. The integration of DR with distributed generation in managing the transmission congestion deduced the overall cost, CO2 emission, and maximum line loading using the Multi-Objective Salp Swarm Algorithm (MOSSA) [46].

For effective scheduling of intelligent home appliances in the energy management system, the Dragon Fly Algorithm (DA) [47], ACO [48], a hybrid of the Harmony Search Algorithm and GWA (HGWA) [49], and a hybrid of Bird Swarm Optimization and Cuckoo Search Optimization (HBCO) [50] are proposed for minimizing the consumer electricity bill, peak to average ratio and waiting time. The dynamic pricing scheme schedules the appliances in off-peak times while considering user comfort. Despite their great performance in solving complex problems, there is no ideal algorithm to deal with any kind of problem, as stated in the No-Free Lunch Theorem by Wolpert and Macready, due to rises concern about the nature of these algorithms that are not adequately mathematics-based, and the convergence is not guaranteed [51]. Recently, the variant of a well-known approach in MA, Particle Swarm Optimization (PSO), has been widely developed to improve its performance. The fact that it can be utilized directly in continuous real number space is one of its advantages. Also, it does not use the gradient of objective function like others. It looks to be a straightforward technique that efficiently optimizes several different functions. Although the PSO was first developed to address unconstrained problems, it has now been used to address constrained problems by utilizing several distinct strategies [50]. Thus, this study aims to overview the PSO-based algorithm in solving the constrained problems in the DSM applications.

The literature survey indicates plenty of review studies on the optimization of DSM. References [52] and [53] has reviewed the application of AI in the energy management system to improve energy efficiency in a smart building. A comprehensive overview of optimizing and controlling the energy system for DR applications in the smart grid, such as smart appliances, EVs, batteries, heating ventilation airconditioning (HVAC) by using reinforcement learning (RL; an agent-based AI algorithm) is presented in [54]. An interesting review in [55] discussed the application of AI in DSM techniques while analyzing its objective function and constraints. However, the limitation of the work only focuses on the residential sector. Scholars in [56] and [57] presented a review on the optimization of DSM in district heating with a few similar goals: profit, renewables deployment, and operation cost. On the other hand, [58] provided a review of the MA based on the energy management system according to different objectives (forecasting, demand management, economic dispatch (ED), and unit commitment). The authors conclude that MA like PSO and machine learning approaches are suitable for forecasting and ED-based application. Moreover, the variety of PSO models for residential load scheduling considering DR has been discussed in [59]. A detailed review of each study is presented in Table 1.

The studies demonstrate that AI modelling in the applications of DSM is a developing research area with opportunities for further review, as each scholar emphasizes the subject



FIGURE 2. Number of publications with respect to years.

differently. While exploring the DSM optimization techniques, we noticed that no study had gathered the research conducted on applying PSO-based models in the DSM and its subfields, such as demand response and energy efficiency. Thus, the authors believe a review within this field could address this research gap and and not duplicate existing work to benefit future researchers. On the other hand, PSO contributes more to swarm intelligence literature than any other SI-based technique in the past years, indicating the significance of PSO and its easiness and practicability in various fields of application [60]. Thus, the PSO models are chosen to support the review work in the DSM application since the abundance availability of research studies related are published in the search engine such as SCOPUS when the following queries are used, which are found in the title, abstract, or keywords.

- Particle AND Swarm AND Optimization AND Demand AND Response
- Particle AND Swarm AND Optimization AND Energy AND Efficiency
- Particle AND Swarm AND Optimization AND Demand AND Side AND Management

Fig 2 depicts the number of publications according to the queries used with respect to years. During these ten years, the research of PSO-based algorithms in DSM applications is rising and become more dominant in the EE than DR programs. Based on this background, this paper aims to systematically review the literature on PSO-based algorithm applications, covering several challenges and opportunities in demand-side management. The contributions of this review paper can be summarized as follows:

• It provides an overview of the theory and implementations of DSM activities considering DR and EE programs

- The application of the PSO-based methods is specifically reviewed in different categories of DSM for different objective functions considering the practical constraints
- It discusses challenges and potential opportunities for investigation of future research areas in the modelling and optimization of DSM

The structure of this paper is organized as follows. Section II presents the overview of PSO. Section III reviews research works on PSO-based algorithm applications in demand response. Section IV emphasizes an overview of PSO-based algorithm application in energy efficiency. Section V discusses the challenge and limitations of the PSO models in the DSM field based on the previous works. Section VI outlines the suggestion for the future direction of research works, and Section VII summarizes the main findings of this paper as a conclusion.

II. OVERVIEW OF PARTICLE SWARM OPTIMIZATION (PSO)

Designing novel computational approaches are usually inspired by a natural and biological system. In the context of MA, natured-inspired algorithms have been utilized for searching and planning, such as finding the sequences of actions required to reach the agent's goals [61]. In particular, swarm intelligence (SI) is a subset of MA featuring the intelligent behavior of biological swarms by the individual's interaction in this environment to solve real-world engineering problems through the simulation of such biological behavior [62]. The Particle Swarm Optimization (PSO) algorithm, which is the major branch of swarm intelligence, is based on the random-search optimization technique [63] and was inspired by social behavior observed in nature, such as schools of fish, a swarm of bees, flocks of birds, and even human social behavior [64]. Each particle in the search space refers to the candidate solution, whereas the food sources represent the global optimum of the problem. The best experience obtained by each PSO particle is called personal best (pBest), and the entire population, known as global best (gBest) throughout the optimization process can be used to modify the search trajectory for that particle. The updated velocity and its position are presented in (1) and (2)respectively.

$$V_{ij}^{t+1} = \omega v_{ij}^t + c_1 r_1 (p_{best,ij}^t - x_{ij}^t) + c_2 r_2 (g_{best,ij}^t)$$
(1)

$$_{ij}^{t+1} = x_{ij}^t + v_{ij}^{t+1}$$
(2)

where t is the repetition number of the algorithm, ω denotes the inertia weight of each agent, v shows the speed of the agent at t (at the i-th row and j-th column), x^t is the position of the agent at t (at the i-th row and j-th column), c_1 and c_2 are the learning factors of the algorithm, r_1 and r_2 denote the accidental numbers, p_{best} is the optimal fitness of each particle and g_{best} is the global of each agent [65]. PSO has been effectively used to resolve a variety of real-world optimization issues because of its simple implementation and promising convergence feature [66], [67], [68], [69]. Since

x

PSO is guided by two mathematical equations which direct the particles to the optimal point, it is much faster than the other heuristic optimization techniques [70]. A few advantages and disadvantages must be considered when applying PSO, as summarized below [71], [72].

Advantages:

- Good memory capability
 - Strong robustness and quick convergence to optimization
- · Has only a few parameters to adjust
- Easier to implement and hybridize with other
- algorithms
 - Computational ability is less influenced by the initial solutions
 - Flexibility in modifying its operators
 - Suitable for optimizing a global search

Disadvantages:

- Easy to trap in the local optima
- Premature convergence
- · Challenging to deal with discrete variables

Numerous works of hybrid and modified versions based on PSO have been introduced since its advent in 1995 to improve the convergence speed and prevent premature convergence [72], [73], [74]. Generally, research on the development of PSO can be classified into three categories: hybrid versions, topological structure, and parameter selection [75]. Table 2 illustrates some of the well-known PSO variants which have been used in past research. However, researchers have shown that PSO variants may have different performances level when solving different problems, thus, it is necessary to find suitable PSO variants for different problems at different stages to efficiently solve the optimization problem [76]. According to the modern intelligent optimization theory, it is essential for an optimization algorithm to keep a balance between exploration and exploitation, an excessive emphasis on one of them will adversely influence another [77]. Thus, the variants of PSO are proposed to strike a balance between exploration and exploitation and overcome the limitations of the conventional PSO.

III. CLASSIFICATION OF DEMAND RESPONSE PROGRAM

DR can be interpreted as the responsive actions by the end-users from their usual consumption pattern to the changes in the electricity price over time or to incentive payments designed to mitigate electricity consumption use when wholesale market prices are high or when the reliability of the system is jeopardized [89]. In other words, DR is the reduction of hourly power consumption in response to high electricity prices [90]. It is noteworthy that involvement in these programs causes the load consumption to be shifted from peak hours to off-peak hours, thus improving the system's adaptability, stability, and dependability [91]. The primary objectives of the DR program include minimizing the total power consumption, reducing the total power generation needed, adjusting demand to match supply availability, and reducing or even eliminating the overloads in the distribution system [92]. Thereby, the issues of peak generation and demand mismatch, peak regulating capacity, and a lack of reserved capacity can be resolved [93].

The research efforts on DR programs can be presented into three methodologies; 1) the control mechanism of the DR procedure, 2) the motivations offered to customers to reduce or shift their consumption, and 3) the DR decision variable [94]. In this sub-section, the discussion is focused on the motivation method offered to consumers for load shifting and lower energy consumption as the initiatives. Fig. 3 illustrates the prominent classes to adjust the electricity load in this method which are the incentive-based DR programs and price-based DR programs. In the incentive-based, the utility incentivizes the consumers to follow the electricity consumption guided from the supply side, meanwhile, in the price-based, the different rates are charged to consumers depending on a certain period, thereby the retail electricity tariff is influenced by the electricity supply cost [92]. The former class is more suitable for the industrial sector, while the price-based is more suitable for the residential sector [95]. Cost-sensitive consumers engaged in the DR programs by modifying energy consumption in response to time-varying prices. The demand response program offered by the utility is crucial to avoid potential system problems such as power imbalances, voltage fluctuations, and blackouts, as well as to save money on capital expenses related to investing in greater generation capacity to satisfy peak load demand [96]. Therefore, a vital need for consumer involvement in the DR program to maintain the supply-demand balance.

A. PRICE BASED-PROGRAM

In this type of DR, the consumers volunteer to adjust their consumption pattern based on the electricity tariffs designed by the utility since the energy cost dynamically varies with time. Time-varying tariffs may be used to utilize the variations in energy prices on the wholesale market directly to customers, causing them to pay for the electricity cost at different hours of the day as opposed to time-invariant tariffs [97]. The pricing scheme under this program is discussed as follows:

1) TIME OF USE (ToU)

TOU tariff aims to modify consumers' daily electricity use patterns. In general, TOU will encourage users to minimize electricity consumption in peak periods and shift it to off-peak periods by adjusting or modifying their electricity usage. In this way, consumers can mitigate the price of their electricity bills by consuming more during the low price and reducing during the high price. Thus, consumers can continue to consume the same amount of electricity while paying a lower electric bill [98]. There are two types of TOU pricing which are static TOU (sTOU) and dynamic TOU (dTOU). The prices in sTOU change by time of day between predetermined price levels and throughout certain time intervals. Seasonal differences may apply. Contrarily, in dTOU, prices

Ref	Optimization techniques		I	OSM aspects		Contributions
	PSO RL	AI	DR	EE	DH	_
[52]		✓		✓		This study grouped the research works on BEMS according to the concept of "Autonomous Cycles of Data Analysis Tasks" while discussing the current challenges and opportunities for each domain.
[53]		✓	~	✓		This study presents the fundamental of a cyber-physical system in a smart building, the role of AI-based approaches in DR and BMS, and summarized the challenges and opportunities for implementation of AI-based applications for smart building
[54]	✓		\checkmark			This study highlighted the interdependence of DR programs, and modelling techniques of various versions of RL as well as the potential control architecture in diverse energy systems considering DG at the building
[55]		~	✓	✓		This study explored various optimization models, technology, infrastructure, communication, and control protocols in RDSM while identifying the challenges in its implementation and future scope
[56]		✓			~	This study presents the terminology and the stages for DSM implementation in DH while analyzing the related research works including simulation and real application in terms of goal, intelligence model, influential parameters in building indoor conditions regulation method and results
[57]		~			~	This study presents the different types of optimization problems, constraints, and techniques in addition to optimization tools used in the district energy system to solve the different district-level problems considering the cooling and heating thermal network
[58]		1		✓		This study presents the research works on AI-based EMS based on five major groups in microgrids while identifying the related problems in forecasting, demand management, economic dispatch, and unit commitment as well as future recommendations
[59]	✓		v			This study evaluated the different PSO methods considering the optimization objectives, constraints, and applications in energy resources such as distributed generation, household devices, heating/cooling devices, energy storage, EVs, and power transformers. The evaluation is based on system design, accuracy, and complexity and discussed in future research.
Proposed work	✓		✓	✓		This study presents a brief overview of DSM implementation while focusing on the different applications of PSO-based algorithms. This study also discusses the challenges and potential opportunities for investigation of future research areas in the modeling and optimization of DSM.

TABLE 1. Comparison analysis of related literature reviews for different review targets of DSM.

change between predetermined price levels, but the timing is not regulated [99]. For example, in Malaysia, a static pricing scheme is adopted for commercial and industrial consumers who are charged based on two periods of peak and off-peak for TOU tariff and the addition of a mid-peak period for a new tariff, Enhanced TOU (ETOU) within a compatible



FIGURE 3. Classification of the DR program.

rate [100]. Policymakers have considered TOU pricing as the feasible DSM option, particularly in regions where smart metering technology adoption has reached (or will reach) double-digit prevalence in the near years. TOU pricing does not need a complicated two-way communication system in terms of smart metering device functionality. This simplifies communication technology used in smart meters as well as data administration for the supplier [101].

2) CRITICAL PEAK PRICING (CPP)

The CPP tariff is an example of an event-based program. A higher rate of energy consumption is charged when critical peaks happen, especially in hours when high electricity demand, thus a higher rate of energy consumption is charged, and consumers are also offered a lower price for the remaining hours (off-hour) [102]. This is due to the inability of utility providers to meet the electricity demand and consequently, the price of electricity will be raised to reduce customer load. If the electricity price stays fixed throughout this peak hour, the demand and supply curves will be imbalanced [103]. CPP specifies the period of the critical peak within an event day, as well as the highest limit of event days annually, but not the exact dates on which the events will take place. As a result, consumers can substantially lower their electricity bill during CPP events through the limitation of load consumption [104]. A case study conducted in [105] showed that residential consumers exhibit a positive effect of demand response in time of CPP event, resulting in a reduction of power consumption. Several advantages of CPP include being easily implemented because it is based on the rate structure of TOU and additionally can enhance the responsiveness of the customer's price by imposing an extremely high rate during critical peak periods [106].

3) REAL TIME PRICING (RTP)

The most complex pricing approach is RTP since the price is established instantaneously according to the supply and demand of the market, resulting in hourly price fluctuations. RTP holds the highest risk with the highest reward to consumers compared to TOU and CPP [107]. Generally, the RTP may be implemented in residential in two ways: (1) as retail pricing signals, and (2) as an integral element of the home energy management systems operation [108]. RTP relies heavily on enabling technology because it must be strongly linked to wholesale market prices and consumer response to allow two side communication. Therefore, enabling technologies such as smart meters are commonly used in RTP to support measurement accuracy [109]. Currently, RTP is more adapted to the electricity supply because of the high penetration of renewable energy like wind and solar into the energy mix, which has caused the balancing of the system to be more complex, thus it would be beneficial to have customer response to wholesale electricity pricing [110].

4) INCLINING BLOCK RATES (IBR)

IBR pricing, as mentioned in the literature, as the demand charge sets a limit on end-user electricity use. If current usage exceeds the stated threshold in a specific period, the consumer will be charged more than the standard rate in that period [111]. In return, the consumers can receive incentives

when distributing the energy consumption at different hours throughout the day to save costs and avoid paying higher electricity prices. Currently, utility companies such as San Diego Gas & Electric, Pacific Gas & Electric, and Southern California Edison offer two-level residential rate structures in which the marginal price in the second level (the higher block) is 80% or higher than the marginal price in the first level (the lower block), depending on the utility. Likewise, the British Columbia Hydro Company in Canada has a two-tier conservation pricing structure, with the second level charged 40% more [112]. A case study conducted in [113] showed that the time-variant tariffs (TOU, CPP, and RTP) give more electricity bill savings compared to timeinvariant tariffs (flat tariff and IBR) with the deployment of PV rooftops. Therefore, the authors in [114] evaluate a combination mechanism of IBR and TOU to mitigate the cost imposed by IBR when the output of PV generation is deficient with batteries.

B. INCENTIVE BASED-PROGRAM

In incentive-based schemes, the utility can monitor and manage end-user appliances and provide financial incentives for

Ref	PSO Variants	Characteristics	Mathematical formulation
[78]	Comprehensive Learning PSO (CLPSO)	Each dimension of a particle in general can learn from different p _{best} for different dimensions for a few	New updating velocity equation: $V_i^d = \omega \cdot v_i^d + c \cdot r_1(Pbest_{fi(d)}^d - x_i^d)$
		generations	where $f_i = [f_i(1), f_i(2), \dots, f_i(D)]$ defines which particles' pbests, the particle <i>i</i> should follow
[79]	Adaptive	The adaptive inertia weight (ω) and the mutation factor $(U^{d}(\sigma))$ are	Updated velocity formula:
	(AMPSO)	combined into the velocity formula to	$V_{k}^{a}(\tau+1) = \omega V_{k}^{a}(\tau) + C_{1}r_{1}\left(p_{k}^{a} - X_{k}^{a}(\tau)\right)$
		have stronger fitting and simulation capabilities	$+ C_2 r_2 \left(p_g^a - X_k^a(\tau) \right) + V_m^a(\tau) \right)$
[80]	Evolutionary PSO (EPSO)	The weights (ω_{ik}^*) of the particle are subjected to mutation while the global	$\omega_{ik}^* = \omega_{i1} + \tau N(0,1)$
		best (b_g) comes randomly distributed	$b_g^* = b_g + \tau' N(0,1)$
			N (0,1) = a random variable with Gaussian distribution with 0 mean and variance 1, τ and τ' = learning parameters
[81]	Cooperative PSO	The vector is split into its component so that n swarms (of s particles) each	$\mathbf{B}(\mathbf{j},\mathbf{z}) = (P_1, \hat{\boldsymbol{y}}, P_2, \hat{\boldsymbol{y}}, \dots, P_{j-1}, \hat{\boldsymbol{y}}, z, P_{j+1}, \hat{\boldsymbol{y}}, \dots, P_n, \hat{\boldsymbol{y}})$
		are optimizing a 1-D vector, instead of having one swarm (of <i>s</i> particles) trying to find the optimal <i>n</i> - dimensional vector	Create and initialize n one-dimensional PSOs: $P_j, j \in [1, .n]$
[82]	Binary PSO (BPSO)	The velocity is regarded as the probability that a position takes the value 1 or 0	The position of each particle = $x_{id} = \begin{cases} 1, rand() < s(v_{id}) \\ 0, otherwise \end{cases}$
			rand() = uniformly distributed random number in [0,1] $s(v_{id}) =$ sigmoid function to transform the velocity to the probability
[83]	Quantum PSO	The state of a particle is shown by wave function $\Psi(x, t)$ of the	$x_i^{k+1} = P + \beta \cdot \left mbest - x_i^k \right \cdot \ln\left(\frac{1}{u}\right) \text{ if } h \ge 0.5$
		Schrodinger equation instead of	$x_i^{k+1} = P - \beta \cdot mbest - x_i^k \cdot \ln\left(\frac{1}{2}\right)$ if $h < 0.5$
		position and velocity	<i>u</i> and <i>h</i> = values generated according to a uniform probability distribution in the range (0,1) β = contraction expansion coefficient mbast = mean of the best positions of all particles
[84]	Quadratic	The introduction of V-shaped transfer	Particle position update:
	BPSO	functions where only the absolute velocity influences the output of the transfer function	$X_{jd} = \begin{cases} \overline{X_{jd}} & \text{if rand } < TF(V_{jd}) \\ X_{jd} & \text{otherwise} \end{cases}$
			$TF(V_{jd}) = \begin{cases} \frac{V_{jd}}{0.5V_{dmax}} \right)^2 & \text{if } V_{jd} < 0.5V_{dmax} \\ 1 & \text{if } V_{jd} \ge 0.5V_{dmax} & \text{if } V_{jd} < 0.5V_{dmax} \end{cases}$
[85]	Speed- constrained	Clerc and Kennedy's constriction factor, χ is adopted to control the	$\chi = 2/(2 - \varphi - \sqrt{\varphi^2 - 4\varphi})$
	Multi-objective PSO (SMOPSO)	particles' velocity which guarantees convergence and avoids large values	Where $\varphi = \begin{cases} C_1 + C_2 \ if \ C_1 + C_2 > 4 \\ 0 \ if \ C_1 + C_2 \le 4 \end{cases}$
[86],	Simulated	The current global best position is	$T_0 = f(p_i)/ln5$
[0/]	(SAPSO)	equilibrium loop of the SA while the	$I_{i+1} = \kappa_T I_i$
		temperature is updated using the adopted cooling schedule in the main	$T_0 = \text{initial temperature} (T_0 > 0)$ $R_T = positive constant is usually taken in the interval of 0.8-$
		loop of PSO	0.999
[88]	Enhanced leader PSO (ELPSO)	Five successive mutation operators are applied to the swarm leader at each iteration. After applying each	First stage: Gaussian mutation $P_{g1}(d) = P_g(d) + (X_{max}(d) - X_{min}(d))$. Gaussian(o,a)ford = 1,2,n
		mutation, if the mutated P_g (global best) has better objective value than the current P_g , it takes the position of the current P_g	Second stage: Cauchy mutation $P_{g2}(d) = P_g(d) + (X_{max}(d) - X_{min}(d))$.Cauchy(o,s)ford = 1,2,n
			Third stage: Opposition-based mutation $P_{g3}(d) = X_{min}(d) + X_{max}(d) - P_g(d)$
			Fourth stage: Opposition-based mutation $P_{g4} = X_{min} + X_{max} - P_g$
			Fifth stage: DE-based mutation operator $P_{g5} = P_g + F(X_e - X_q)$

TABLE 2. List of some PSO variants with their characteristic and mathematical formulation.

peak-hour demand reduction to engaged customers, and consumers receive a discount rate for their participation. The following are the main scheme under this program:

1) DIRECT LOAD CONTROL (DLC)

DLC enables utility companies to regulate consumer demand remotely by rescheduling or switching on and off certain household appliances. In exchange for the inconvenience caused to the consumer, the utility provides an incentive payment or credit. As such, utilities may control lighting, thermal comfort equipment, refrigerators, and pumps. This includes benefits such as more precise estimation, deep monitor for loads that can be reduced during peak periods, and creating effortlessly simple DSM for customers [115]. A case study of residential DLC control mechanism on air- conditioning (AC) has been investigated in [116] which has given impacts on the demand and environment side through reductions of peak load and CO₂ emission. Nonetheless, issues about a large number of devices and customer security have been significant obstacles to the introduction of DLC [117]. To overcome this problem, the authors in [118] proposed an active database for the DLC system to control automatically the on and off status of the consumer load without the intervention of the operator.

2) INTERRUPTIBLE/CURTAILMENT LOADS

The option for curtailing load during system outages is included in retail tariffs with a discount rate or bill credit by lowering the load consumption. Failure to curb may result in penalties. Usually, interruptible programs were only available to the largest industrial (or commercial) users. However, industrial customers with continuous processes might not be suitable for this program [119]. The least sizes of customers required for interruptible/reduced rates for the standard interruptible program range from 200 kW to 3 MW, depending on the condition and market. Under these tariffs, consumers commit to either reduce the consumption to a predetermined threshold or reduce certain blocks of electricity load. Normally they are required to curtail between 30-60 minutes after being informed by the utility, which is typically done through AMI. However, there is a limitation on the number of times or hours for the utility to request an interruption (less than 200 hours annually) [120]. Therefore, by lowering its peak demand, the utility saves money on expensive power reserves, improves service quality, and ensures reliability. Customers benefit from lower energy costs and incentives offered by the ISO or the local utility [121].

3) EMERGENCY DR PROGRAMS (EDR)

Emergency DR systems offer consumers incentive payments for lowering their demand voluntarily when durability- triggered events occur. However, consumers have the option of foregoing payment and not curtailing when notified. Furthermore, no reduction in consumers for their load consumption will not be punished. Typically, the amount of the incentive rewards is determined in advance [122]. In the case of the wholesale market, the Independent System Operator (ISO) will adopt EDR programs to lower peak demand and prevent price spikes [123]. A big number of customers voluntarily participate in the EDR program in response to ISO's announcement [124]. As an incentive, the customers will receive a huge amount of money as a payment of approximately 10 times the off-peak electricity price by ISO [125]. For example, in the United States, the New York Independent System Operator (NYISO) provides two EDR programs which are NYISO-EDRP and NYISO-SCR (special case resource) to respond to NYISO operating instructions by reducing load when operating reserves are expected in low conditions or during an emergency of a system [126].

4) DEMAND BIDDING PROGRAMS

In DBP, the opportunity is given to consumers by the electricity trading markets to select a time and way for real-time and day-ahead spot market participation. Thus, by removing the load, the consumer will receive a market price when the market operator requests it, similar to the payment for generators to supply. Customers will negotiate on a certain price decrease, timeline, and availability, and their offers will be filtered and chosen according to market demand. In most cases, for the highest accepted bid offer or for developing demand-side bidding markets, a minimum fixed rate will be paid to all bidders [126]. For example, the Quick Bidding Program (QBP) is proposed by [127] in Kuwait with features that the targeted reduction load by the utilities will be displayed on their websites within the time frame. Consumers can quote their capability to lower the load for different periods. Meanwhile, a practical example of this bidding mechanism is the NYISO's Day-Ahead Demand Response Program (DADRP) which allows the consumers to bid their load reductions into the Day- Ahead energy markets as generators do. The payment of offers that are determined to be economic is paid at the market clearing price [128].

C. PSO-BASED APPLICATIONS IN DEMAND RESPONSE

The main attributes of each relevant research work are presented in Table 3. PSO is an effective way of solving largescale non-linear optimization problems [129]. The reason for practicing PSO in DR optimization is because of the tendency to give impactful and accurate results [130]. It can quickly locate a near-optimal solution while requiring less effort than other mathematical methods for solving a non-linear optimization problem in DR. Moreover, PSO is the most commonly used optimization algorithm for solving DR optimization problems [94], [131]. Thus, it is proposed to address the DR management problem in the present works. From the observation, the authors mostly investigate the working of PSO models in the different DR programs based on one or more optimization objectives: (1) energy or electricity cost minimization, (2) peak-to-average ratio (PAR) minimization, (3) peak load reduction, (4) maximize operation profit and

(5) maximize the use of renewable energy resources (RER) while considering user comfort. This is important to attract the consumer to respond or participate in DR programs when several constraints such as temperature regulation, optimal state of charge (SOC) for EV or energy storage system (ESS), and favorable timing for appliance use are applied. Moreover, the integration of RERs such as wind and photovoltaic (PV) solar with ESS, EV, and the battery has been studied in most of the research works to achieve those objectives when the charging is during off-peak hours while discharging is during peak hour in response to DR pricing or incentives.

IV. CLASSIFICATION OF ENERGY EFFICIENCY

A. ENERGY EFFICIENCY IN BUILDINGS

Energy efficiency (EE) can be defined as a long-term conservation strategy that aims to save energy and reduce demand through energy-efficient processes [155]. Energy efficiency gives huge benefits such as the reduction in the required number of energy resources to achieve a specific amount of energy service, along with associated effects on depletion of resources, energy safety, and cost-saving; as well as the decrease of carbon emissions, other pollutant emissions, and overall environmental impact linked with electricity consumption [156]. According to the Efficient World Scenario (EWS), energy efficiency may lower yearly energy-related emissions by 3.5 GtCO2-eq (12%) based on 2017 levels, accounting for reductions greater than 40% of the reductions necessary to comply with the Paris Agreement [157]. Therefore, the combination of energy efficiency with renewable energy and other measures is critical to achieving the target of global climate. Generally, EE practices in buildings consist of 1) active measures: optimizing the HVAC system, energyefficient appliances, and lighting, renewable energy utilization, and managing the energy effectively with regards to occupant's comfort and 2) passive measures: lowering energy consumption by utilizing the potential of nature's lighting, cooling, and heating [158]. The EE in buildings is determined by several factors such as the degree of electrification, the level of industrialization, the amount of building area per capita, the existing climate, and policies at a local and national level to promote energy efficiency [159].

Another action under the EE strategy is energy conservation (EC) which focuses on changing the behavior of people to utilize energy more efficiently. In the energy pyramid, EC is the first step to achieving sustainable energy as it is located at the base with the least cost option [160]. The conservation of energy in residential can be implemented either by changing the consumption of energy services or spending on energy-efficient appliances [161]. In general, EC behaviors are continual and repetitive actions to reduce consumption daily that requires compromising comforts or reducing economic utility to save energy driven by several factors such as social-psychological and environmental concerns [162]. Energy conservation measures (ECM) in buildings are classified into three types: major investments, minor investments, and zero investments. ECMs have recently drawn increased attention due to their useful application in both newly constructed and existing structures. Initially, the potential of energy-saving for ECMs is assessed by simulations, and then the appropriate ECMs are chosen for implementation in actual buildings [163]. Several passive measures, such as insulation on a residential house [164] show significant energy-saving potential and collectively reduce the energy performance index (EPI) by 34%. The insulation of a building may maintain the cool or heat of the house internally while restraining heat flux with the surroundings since the thermal insulator material can reduce the rate of heat flow [165]. In comparison, the active measures taken in [166] could reduce energy by 63.5% in EPI, which is significant savings obtained through the replacement of ordinary appliances with energy-efficient appliances. Moreover, the modelling results in [167] show that the most effective way to reduce evening peak demands is by switching light bulbs to LED, which results in reductions of total appliance electricity demand by 18.8%, total residential electricity demand by 14.2%, and total national electricity demand by 5%. Whereas authors in [168] adopted both active and passive measures in buildings and discovered that installing PV systems and decreasing lighting power density had the best energy-saving ratio.

B. MEASUREMENT AND VERIFICATIONS (M&V)

Measurement and verification (M&V) are necessary to test the performance of each energy conservation measure to ensure its efficient implementation and operation. M&V is a process of employing measurements to accurately identify the energy savings achieved by an energy efficiency intervention in a specific building or facility. The energy savings reflect the absence of energy consumption, so they cannot be analyzed instantly, thus, the evaluation is usually made through a comparison of energy consumption at the facility before and after the adoption of a retrofit measure, keeping any changes in circumstances [169]. The major purpose of M&V is maximizing the accuracy of energy savings, optimizing financial efficiency projects, increasing the public understanding of energy management, and addressing the significance of emission-reduction credit [170]. Part of the M&V methodologies to be followed is fitting and maintenance of meter calibration, collection and analysis of data and justifiable results, and verification of reports [171]. The International Performance Measurement & Verification Protocol (IPMVP) defines four different M&V options: partially measured retrofit isolation (A), retrofit isolation (B), whole facility (C), and calibrated simulation (D) [172].

In any M&V project, three periods occur sequentially, consisting of the baseline, implementation, and reporting periods. The engineering or statistical methods are usually performed to estimate the adjusted baseline in the reporting period by normalization calculation [173]. Therefore, it is important to maintain its accuracy and minimized the uncertainty throughout the process to determine its success.

TABLE 3. Research works on PSO-based applications in the DR program.

Ref	DR Program	PSO-based method	Objective function	Constraints	Key Findings
[132]	RTP	Binary PSO (BPSO)	To minimize the electricity cost, system PAR and user's discomfort with the integration of PV, ESS and EV in HEMS	The charging and discharging of ESS and EV based on RTP pricing, current, and maximum SOC, power demand limit and availability of EV	Cost reduction by 52.47% from the base schedule price, while the PAR and DI were reduced by 15.11% and 16.67% respectively through optimal scheduling by BPSO
[133]	RTP	Binary PSO (BPSO)	To minimize the electricity bill for prosumers and consumers (PV & wind, and EV) through Peer to Peer (P2P) energy trading in the smart home	Power balance, appliances operation, the charging and discharging of battery and EV	The electricity cost minimization is achieved through optimal scheduling of appliances and utilizing P2P energy trading when the excess power of DG is traded to consumers instead of the grid
[134]	TOU	Hybrid PSO	To minimize the energy cost and total tardiness in the flexible flow shop scheduling for the multi-objective optimization	Completion time and job processing	The total tardiness and the electric power cost are minimized simultaneously while maintaining production efficiency by reducing the machine speed and higher electricity consumption at off- peak times
[135]	CPP and RTP	Multi-objective PSO (MOPSO)	Reduction of energy cost and PAR while maximizing user comfort in HEMS with the integration of RES	Operating time of shiftable devices and their usage	RTP gives greater electricity cost savings compared to CPP
[136]	TOU and developing TOU (D-TOU)	PSO	Reduction in building peak load and customer electricity cost with optimized thermal energy storage (TES)	TES system capacity and its operation	Load shifting and reduction of peak load for utility and cost reductions obtained by a customer with an optimally configured TES system
[137]	CPP, DLC, TOU, RTP, Interruptible	PSO	To minimize the generating costs along with the operating cost considered DR programs- based unit commitment	Price elasticity of demand and welfare functions of customers	CPP and DLC give the maximum reduction of peak demand while TOU shows the highest reduction in the total cost
[138]	CPP, TOU	PSO	To maximize the profit in manufacturing considering the electricity and labor cost	Production target	During summer, the load curve of profit under CPP is greater than the load curve of profit under TOU
[139]	TOU	Multi-sub swarms PSO (MSPSO)	To minimize the monthly electricity charge for commercial building	Particle velocity and particle position	As compared to PSO, MSPSO could save approximately 17% on monthly charges when peak load is reduced during on-peak periods in the scheduling simulation
[140]	TOU	Binary PSO (BPSO)	To find the best sizing and placement for electrical energy storage devices (EESDs) on an improved IEEE 24 bus test system in smart grids	Sizes of the EESDs and the budget for investments in EESDs	Load factor improvement and cost reduction with DR program associated with EESDs since during the load demand is reduced and shifted to valley periods.
[141]	CPP, RTP, TOU	PSO	Minimization of cost for operation, pollutant treatment, and carbon emission in household energy management under different RER and EV	Distributed energy resources sizing, power balance, distributed generation power limits, exchanging power limits, EV charging and discharging limit, ramp rate limit, and line transmission capacity	The overall result shows that the proposed approach achieved the objectives and provide a good solution while maximizing RERs and EVs integration under DR programs during summer and winter

[142]	Enhanced TOU (ETOU)	Evolutionary PSO (EPSO)	To reduce the electricity cost for electronic manufacturing facilities through simultaneous load management strategies	Total energy consumption before and after the optimization should not be more than 5%	Reduction of monthly electricity cost gain when the maximum demand value located in the mid-peak zone and peak demand value was decreased
[143]	Curtailment load	Quadratic PSO (QPSO)	To solve the unit commitment in DR with the integration of microgrid (MG) considering the uncertainties	Minimum spinning reserve, operational of MG	The DR integration into MG reduced its operation cost by 15.31%
[144]	EDR, Interruptible, DLC, TOU	PSO	To minimize the cost of purchasing and energy supply production with the integration of RES as well as maximization of financial profit for microgrid	The total power produced by RER, the main grid, and the generator must be equal to the sum of power needed for two consumers	The energy reduction, total compensation, and incentive savings achieved due to load shifting from peak to off-peak periods
[145]	DLC	Binary PSO (BPSO)	To increase the temperature of the water heater over 24 hours while reducing the total peak load demand for the residential load	The weight factor for temperature is bigger than the total load weight for customer satisfaction	Peak load reduction for about 500W-750W per household and obtained potential cost savings for utility and household user
[146]	Interruptible	Binary PSO (BPSO)	To meet a requirement of a system for total curtailments in an hour while minimizing both total payment and the frequency of interruptions imposed on customers	Operational of the available interruptible loads	BPSO can solve a multi- objective optimization problem, minimizing objective functions, and the number of interruptions is reduced and distributed across all loads
[147]	Interruptible	Improved Cooperative PSO (ICPSO)	To optimize the profits for the operation of a virtual power plant (VPP) and VPP model of energy management while maintaining the comfort of the occupant	Conventional unit, ESS charging and discharging, Flexible load adjustment, energy supply, and demand matching	Increased operation profit by 10.82 and 2.38%, respectively, in comparison to the optimization approach that solely took into account interruptible and rigid load
[148]	RTP	Binary PSO (BPSO)	Minimization of energy cost, carbon emission, PAR, and occupant discomfort considering RESs and ESSs	Operation of residential appliances, charging/discharging of ESS	Decrement in cost (2.89%), PAR (18.75%) carbon emission (14.9%) compared to unscheduled with the presence of RESs and ESSs
[149]	TOU	PSO and Improved PSO (IPSO)	To reduce the total electricity cost and PAR integrated with Plug-in Electric Vehicle (PEV) through an effective approach of sharing economic scheme in a community	The amount of selling power, batteries operation, and the total energy supplied less or equal to the power consumption of the building	Reduction by 23% of electricity cost for each home and 37% of peak demand and PAR for the total community
[150]	Interruptible	Multi-objective variants of PSO (MOPSO)	To optimize the size of a hybrid system consisting of PV and diesel generator through load variation factor (LVF)	Maximum number of system components and maximum value of LVF	An affordable and dependable hybrid energy system consists of a PV panel and diesel generator with optimal load reduction portion
[151]	EDR	PSO	To optimize the EDR model of central air-conditioning (CAC) simulating consumers' trade- off between thermal comfort levels (TCLs) and profits	The indoor temperature at the initial time and the preference of consumers	The increase in the penalty prices is more significant compared to the increase in compensation prices for the EDR level of CAC improvement
[152]	EDR	PSO	To coordinate convenience and economy by planning the energy use of the user's indoor equipment in a HEMS with scattered generations	The working characteristics of appliances, user preferences, charging/discharging power of energy storage equipment	Under different EDR incentives, user-obtained load reduction and cost benefits prove the solution scheme can solve home energy scheduling problems under uncertain environment

TABLE 3. (Continued.) Research works on PSO-based applications in the DR program.

[153]	TOU, RTP, Curtailment load	Enhanced Leader PSO (ELPSO)	Optimal scheduling of HEMS & bill reduction	The range for starting time of each appliance, time intervals for appliances to ON in a day	 Greater reduction of consumer's electricity bills while maintaining their convenience ELPSO outperforms other basic models
[154]	TOU	Simulated annealing PSO (SAPSO)	To reduce the peak load under TOU pricing optimization	User's benefit, benefit to the power supplier, electricity rate, marginal cost price, electricity consumption similarity, total electricity consumption	The proposed method maintains the satisfaction level and power consumption of consumers while decreasing peak load and increasing valley load.

TABLE 3. (Continued.) Research works on PSO-based applications in the DR program.

Measurement, sampling, and modelling are the three key uncertainty elements to consider when reporting savings with reasonable statistical precision [174]. The acceptable levels of uncertainty are determined by the point at which savings are greater than twice the baseline model's standard error, according to the IPMVP, which offers a systematic methodology for assessing each element of uncertainty added to a project [175].

The simulation-based method in Option D necessitates model calibration using measured data in an hour or month. The recommended process to verify the whole building or specific building components for its performance and verification used is calibration simulation, which is included in Option D [176]. Building energy models are often calibrated by incrementally changing parameters of the model from audit data until predicted energy use is within specified tolerances from actual measurements. The parameters that have the greatest influence on the major indicators of building energy performance predicted by the model are often identified through uncertainty and sensitivity analyses, and during the calibration process, these parameters receive heavier weightage [177]. The acceptable tolerances for the validation of building energy models are outlined by IPMVP [178] and ASHRAE Guide 14 [179]. Nowadays, automated calibration has gained interest compared to conventional methods due to its faster and more efficient processes such as Bayesian calibration [180], pattern-based calibration [181], and multistage calibration [182]. Furthermore, the revolution of M&V 2.0 has offered new technologies that might lower the cost of M&V, generate quicker findings with greater transparency and confidence, and thus raise the acceptability of the savings projections [183]. More advanced technologies of M&V 2.0 tools that have been commercially available in the market have been discussed in [184].

C. ENERGY MANAGEMENT SYSTEM (EMS)

The definition of EMS is a computer system consisting of a software platform for basic support services and a collection of applications for the functionality required to ensure sufficient energy supply security with minimal price and efficient operation of electrical generation and transmission facilities [185]. The purpose of EMS is to optimally distribute various energy sources to consumers while integrating renewable energy sources without jeopardizing the system's dependability, security, or safety [186]. The benefits of EMS include low operational costs the privacy of consumers, diversifications, and a less computational load [187]. In the smart grid (SG), the main energy management objectives including EE, demand profile enhancement, cost optimization, and consumer comfort. SG delivers energy more efficiently, improving customer utility interaction, and modern management techniques, and responds to wide-ranged events occurring in the system [188]. Technologies such as human-machine interfaces (HMI) and supervisory, control, and data acquisition (SCADA) are helpful for effective EMS implementation, especially if a certain demand is supplied by more than one energy source. The EMS strategy can be classified into a rule-based approach and an optimization approach as shown in Fig. 4 [189]. The first approach manages the power demand by implementing a fixed rule based on the efficiency maps of the equipment, whereas the latter approach uses the common strategies which are equivalent consumption minimization strategies (ECMS) and model predictive control-based strategies (MPC) [190]. PSO can be used to optimize the structures and parameters of these methods. The current literature reviews the various control and operation strategies of EMS in hybrid renewable energy [191], microgrid [192], and hybrid electric vehicle [193] applications integrated with optimization techniques to have better performance and achieve some objectives.

Meanwhile, the contribution of the home energy management system (HEMS) in the residential sector has been promoted in references [194] and [195] to coordinate and schedule the home appliances according to certain criteria, thus improving the energy efficiency and demand flexibility of the buildings. The several functionalities of HEMS including providing a detailed overview of the graphical data of energy consumption have a few advanced functions for monitoring, information, and automation purposes and to forecast the loads and local generations at the household levels [196]. IoT technologies have been incorporated with HEMS during the past several years and are essential for the intelligent control and management of the system's end users.



FIGURE 4. EMS strategies classification.

Furthermore, the AI and ML technologies that can be trained and used to forecast the near future are implemented into modern smart houses [197]. A recent study of consumer willingness to adopt HEMS was analyzed in [198] based on different interactions such as technology attributes, attitudes, and infrastructure.

D. PSO-BASED APPLICATIONS IN ENERGY EFFICIENCY

Table 4 presents the research works on energy efficiency with different energy systems. In terms of optimization models, most researchers studied optimizing energy systems and improving energy efficiency by using single or multi-objective PSO models. Without being restricted by the burden of dimensionality, PSO can quickly solve extremely challenging constrained optimization problems [188]. The integration of EE measures with RERs is a promising mechanism that can effectively drive the low-carbon energy transition under practical constraints. Meanwhile, the optimization based on EMS focuses on achieving global optimum by minimizing cost functions such as energy consumption and total cost. Besides, optimization-based calibration has been adopted in the studies and uses accuracy metrics for accuracy measures such as mean average error (MAE), mean bias error (MBE), and coefficient of variation of root mean square error (CV(RMSE)).

V. PSO-BASED ALGORITHMS IN DSM: CHALLENGES AND POTENTIAL SOLUTIONS

The key challenges of implementing PSO in DSM are categorized based on the complexity and uncertainty of the modelling and optimization. The complexity in the modelling is addressed for multi-objective optimization problems considering the large number of parameters used, the system architecture, and constraints. Additionally, due to uncertainties in the optimization process, it is quite challenging for researchers to develop a stochastic model.

A. COMPLEXITY IN MODELING

The complexity of the optimization process directly influences the convergence speed and accuracy of modelling outputs [58]. In the previous work reviews, PSO applications in DR and EE fields show the intermittent penetration of RERs, ESS, EV, and multiple household appliances in load scheduling while considering user preferences. Various PSO-based models are used to determine the optimal schedule for devices to make the scheduling process simpler, however, there is a trade-off between optimality and complexity [222]. The high modelling complexity has increased the risk of infeasibilities or slow convergence. The consideration of multiple classes of load scheduling such as shift, fixed, and interruptible appliances in HEMS caused the longest waiting times for the appliances to shift to the times desired by the consumer [135] and longer convergence times [223], thus increasing user dissatisfaction. In the smart grid, the interruptible loads are bounded by the time duration specified by users, leading to a vast number of potential solutions. Thus, the swarm formulation needs to modify ensuring the solutions are within the user-specified time frames [224]. For large-scale situations, the original PSO model has longer computation times in DSM problems involving PEV charging and discharging [149] and PV systems [225] considering multiple homes in the residential community. Large population sizes and iterations require additional computation and increase both computation time and the algorithm's reliability [226]. In addition, the privacy

IEEE Access

TABLE 4. Research works on PSO-based applications in the EE program.

Ref	EE Program	PSO-based method	Objective functions	Constraints	Key findings
[199]	M&V	PSO	To construct accurate baseline models	Data range	The suggested approach is estimated to save M&V costs and time while increasing the calculation's accuracy for energy performance.
[200]	Calibration	PSO	To optimize the existing manual calibration and modify it into an automated process	Input parameters	The automated optimization calibration improves the accuracy and efficiency of the calibration
[201]	Multi-stage building calibration	PSO and hybrid Generalized Pattern Search PSO with Constriction Coefficient Hooke-Jeeves (GPSPSOCCHJ)	To calibrate the different sub-models separately and obtain an accurate model	Calibration error limits	Hourly temperature data had a lower MBE and CV(RMSE) of -3.74 percent and 6.03 percent, respectively
[202]	Fuel cell	Adaptive mutation PSO	To reduce energy consumption and improve output stability	Dynamic response and protection elements	The improvement of lithium battery charging and discharging rationality and fuel cell stability
[203]	Heat pump integrated water heater	Multi-objective PSO	To determine the ideal operating point for the air flow rate and compressor speed	Size of household appliances	Reduction of energy consumption by 16% with a minor increase in noise level and operating time
[204]	Building envelope and air-conditioner	PSO	To minimize annual cooling energy consumption for a residential building	Not specify	 Reduction of annual cooling energy consumption up to 53.78% compared to baseline PSO is more profitable compared to GA for the simulation of the building optimization process
[205]	Heating load (HL) and cooling load (CL) of HVAC	PSO-Multi-Layer Perceptron (PSO-MLP)	To reduce the HL and CL of residential energy- efficient buildings	Not specify	 The performance of MLP is more efficient with a decrease in MAE and RSME for both load PSO shows more robustness compared to ABC in optimizing the parameters of MILP
[206]	Operation system	Renewal PSO (RPSO)	To boost system profitability and energy efficiency	Interactions between machines and buffers	The proposed control method showed resilient operation against random disruption events and adapted well to time-variant system reliabilities
[207]	Cooling water system	Hybrid programming- PSO (HP-PSO)	Enhancing the cooling water system's performance	Limits on the cooling tower's outlet water temperature, the range of control for the cooling water pump and cooling tower fan, and the variety of electrical devices	Potential for a 15.3% reduction in the energy used by the cooling water system.
[208]	Building calibration	PSO	To generate lumped parameter-building models with similar dynamics for different ECMs.	Not specify	Models of retrofitted buildings can be calibrated using the suggested approaches with an acceptable degree of accuracy (MAE $\leq 0.5^{\circ}$ C).

[209]	Lighting devices	Distributive-PSO (DPSO)	To minimize power and produce sufficient illuminance	Dimming instructions and minimum illuminance	The suggested technique effectively controls luminance in actual for a range of working environments.
[210]	Plug-in hybrid electric vehicle (PHEV)	Multi-objective PSO	To reduce the driver's action on the steering angle, minimize the mass of electrical components, fuel consumption, and exhaust emission	The angular speed of the ICE, gearbox neutral position, RMSE value of the optimized vehicle, PHEV consumption	Reduction of the driver steering action by 71.9%, reduction fuel consumption by 18.44% and reduce pollutant emissions by 69.34% of CO, 20.33% of HC, and 22.11% of NO _x
[211]	Lighting	Multi-objective PSO	To enhance the visual comfort of users and energy savings in a building	Luminaire's dimming average, illuminance level limit, illuminance uniformity	Energy efficiency and illuminance uniformity improved by 58% and 17% respectively compared to the baseline
[212]	Sewage treatment system	PSO	To improve energy efficiency	Not specify	Achieved a good energy- saving effect
[213]	Internal combustion engine	Improved mutation PSO (M-PSO)	Minimization of total cost and CO ₂ emission	Energy balance, components operational	The varying weight ratio of the two objectives has a greater impact on the annual total cost-saving rate compared to the annual CO ₂ emission reduction rate
[214]	EMS for ship hybrid energy system	Quantum PSO (QPSO)	Low operating costs and optimal use of clean energy for ship electric power systems	Clean energy power generation, energy storage power supply, fuel cell power generation, diesel power generation, and load power balance	 Reduction of operating costs, fuel costs, and pollutant emissions of marine power system Achieved the environmental requirements of the Energy Efficiency Operating Index (EEOI) for ships
[215]	EMS strategy of hybrid solar-hydro system	PSO	Size optimization based on the levelized cost of energy (LCOE) minimization	The water level in the upper and lower reservoir	Configuration with hybrid PV-battery integrated with pump hydro-storage system has greater power system efficiency and more reduction in LCOE compared to another model
[216]	EMS strategy for PV system and BESS	PSO	Finding the best PV and BESS size given the specified energy cost	SoC of BESS and system operation	The suggested approach can find the most cost-effective sizes for PV and BESS
[217]	EMS of microgrid for virtual power plant (VPP)	Multi-objective feasibility enhanced PSO (MOFEPSO)	To reduce the total operating cost and net emission	Power balance, storage limits	The proposed method reduces the operating cost, and GHG emissions and optimized the scheduling operation of microgrid for VPP
[218]	EMS for fuel cell hybrid power system	PSO	To make the proton exchange membrane fuel cell (PEMFC) hybrid welding robot system more stable for real-time performance and economical	SoC of lithium battery, fuel cell output power	The stability of fuel cell power output is improved by 11.26% and reduction of hydrogen consumption by 3.24%
[219]	HVAC	PSO	To propose a new approach to optimizing HVAC energy consumption based on a prediction of load and energy flexibility	Prediction of chiller loads	The discounted payback value is 5.8 years, and the energy-saving ratio is roughly 10%
[220]	EMS for microgrid of building prosumers	PSO	To minimize the operation cost in the building integrated with RES, ESS, and PEVs	Operation and technical of the buildings	 Achieved cost savings in the range of 11% Lower nominal power of the HVAC system can be used
[221]	HVAC	Artificial Neural Network (ANN)-PSO	To ensure optimal operation of the HVAC system and control indoor environmental parameters	Indoor air temperature, indoor air humidity, indoor air CO_2 , indoor air volatile (VOC)	7% overall energy reduction without deteriorating the conditions of the internal environment

TABLE 4. (Continued.) Research works on PSO-based applications in the EE program.

issues of individual homes in the community should be a concern for the optimization process where the system design should not collect power usage profiles of all users in centralized ways. Thus, the practical optimization model must be modelled concerning privacy issues and other parameters like cost and carbon emission [227].

The standard PSO is only suitable for the optimization problems in continuous space, thus its variant model for a discrete domain such as discrete PSO (DPSO) and binary PSO (BPSO) are proposed for the combinatorial optimization problems in discrete space, in which the trajectories of particles are defined as the changes in the probability and the velocity is transformed from real number space to probability space via a sigmoid function. Moreover, HEMSbased scheduling often required binary formulation for the ON and OFF status of the appliances and a large number of decision factors, thus the optimization problem would be more difficult than the problem with integer decision variables [84]. The DPSO algorithm is able of creating adequate diversity during the search, although it frequently suffers from the disadvantage of being trapped in local optima [134]. Thereby a hybridization of variant PSO models such as hybrid BPSO and GA (HBPSO) [228] and hybrid DPSO [134] are proposed for more complex problems. The multi-objective optimization which is more complex than the single optimization problem often involves constraints and user preferences, thus the solution to the problem is a challenging task. Considering user comfort would lead to an increase in cost and PAR when optimizing the three parameters together in a multi-objective problem [229]. Therefore, requires a search for the best trade-off between these objectives by adjusting the weighting coefficients in different cases to find the best possible solution [132].

VPP energy management is similarly a challenging task because of the coordinated functioning of various energy supplies and the associated uncertainties which requires an optimal algorithm to ensure smooth and reliable operation in real-time without sacrificing the optimal operation costs. A basic PSO model suffers from early stagnation and loses exploration capabilities during the latter evolution stage when addressing the complicated issue, thus, a hybrid PSO may be developed to minimize the iteration number. PSO model suffers from premature stagnation and loses exploration ability during the later evolution period while solving the complex problem, thus a hybrid PSO can be utilized to lesser the iterations number when real-time data is used [230]. Furthermore, when basic PSO integrated with a time-varying acceleration coefficient (PSO-TVAC) dealing with complex problems such as optimization of the operation and schedule of air conditioning system under different conditions, it is very difficult to search for a feasible solution directed to more trouble if the scheduling constraints are considered in the early stage of iteration [231]. Thus, the modified PSO (MPSO-TVAC) method is introduced to increase the exploration ability and rate of success for a global optimum by including a "random viable solution" into the standard PSO-TVAC.

B. UNCERTAINTY IN OPTIMIZATION

In the optimization process, the consideration of uncertainty directly reflects the uncertainty in real-life problems, which might impact model results and the optimum value of the objective function. Therefore, any global optimization search includes an addition element of uncertainty [232]. The motivation to develop stochastic modeling in DSM stems from the challenge of confronting the uncertainty and fluctuation of RERs, dynamic pricing of tariffs, environmental variables, and random user behavior. The issue with stochastic variables is their deterministic constraints. Under such conditions, uncertainty is diminished to a certain extent [233]. Thus, robust optimization has shown to be a promising technique for addressing the uncertainties in the optimization issue problem. Authors in [234] consider the uncertainty of production from wind and PV panels to solve the day-ahead energy management in buildings. In order to introduce higher exploratory properties in the search procedure, the fixed parameters in the traditional MOPSO are modified into a mutation of the strategic parameters in EPSO. This adjustment increased the cover rate and the overall front of the non-dominated solutions. In view of the shortcomings of PSO such as lack of randomness in particle position changes and numerous parameters, the theory of quantum mechanics is combined with PSO to solve the non-linear and non-convex optimization problem considering high uncertainty from the power output of PV and wind turbine in microgrid [235]. The optimal scheduling method for power resources in microgrid considering the uncertainty of renewable energy output is vital especially when the generation of actual maximum power from wind and PV is less than the power arranged in the scheduling plan, increasing operating costs and frequency fluctuation of the microgrid which may harm the system [236]. Moreover, the imprecise prediction of RERs generation output directly led to non-optimal energy management and programming, and increase households' electricity costs, thereby restricting the benefits of smart homes [237].

Dynamic pricing schemes like RTP has the potential to reward consumer fully but it also has the potential to maximize risk when they are unable to manage the use of electricity prices depending on the predetermined interval [238]. Due to hourly price changes, RTP flexibility may reflect load patterns or generating costs. RTP is more flexible than both TOU and CPP, but it has the disadvantage of consolidating numerous appliances in low-cost energy zones. To handle it, the author in [239] presented a system by combining IBR and RTP to adjust energy price rates during the low energy price period depending on appliance power consumption. When the overlapping time of appliances is customized with IBR, the fitness function of PSO integrates a modified IBR to

TABLE 5. Opportunities for new application under DSM.

Ref	New application	Suggested PSO-based method
[242]	Integrating EVs along with distributed generation into the energy management problem	MPSO
[243]	Optimizing the CO_2 and $PM_{2.5}$ concentrations simultaneously in the real building	Back-propagation neural network combined with adaptive multi-objective PSO (BPNN-AMOPSO) based on computational fluid dynamics
[244]	 Trigeneration system of electricity, cold and heat Cold applications such as refrigeration or deep-freezing 	Not specified
[245]	Integration for load forecasting results for individual end users to observe the overall forecast accuracy for a community and the effect of combining load forecasts and HEMS for the whole community	Not specified
[246]	Combination of energy management strategy and the flat controller by including the controller parameter in optimizer as optimization variables	Not specified
[136]	The determination of optimal electricity prices for industrial loads based on dividing weekdays into smaller periods, namely super off-peak, off- peak, peak, on peak and super on the peak in conjunction with partial and full storage capacities for thermal energy storage system	Not specified
[247]	The integration of electrical energy storage and the price of real estate from the perspective of a win-win situation between consumers and investors	Not specified
[248]	The steering system energy savings with the influences of the steering assist characteristic and design of the variable assist characteristic curve based on maneuverability and energy savings	Shuffled PSO (SPSO)
[249]	Economic dispatch problem associated with power system operations as well as the convergence characteristic	Double-weighted PSO (DWPSO)
[250]	 The optimization of peak grid load more directly and grid carbon intensity reductions by optimizing costs and the system's carbon footprint in combination Energy sharing in smart, connected communities with energy storage systems Detailed modelling of the techno-economic feasibility of different battery chemistries (including storage capacity in EV) Assigning the use of the different zones in the building to different user types based on their specific thermal preference 	Not specified
[251]	 The energy consumption characteristic of the unit coupled with a heat pump or heat storage tank Optimization of thermal power load distribution among the modified units based on the energy consumption characteristics 	Momentum PSO (MPSO)
[252]	Creating a generalized model feasible for other locations and layouts in the country and beyond for the hybrid renewable energy systems	Not specified
[253]	Different types of liquid air energy storage systems	Not specified
[254]	 Comprehensive analysis of evaluating self-discharge as a parameter varying with temperature in the energy storage system Considering the effect of environmental parameters on the design of the configurations as optimization objectives Pump hydro storage operation for switching between two energy storages by varying the proposed parameters 	Not specified
[255]	Multi-generation energy systems with different energy technologies to any case studies such as buildings, universities, and district	PSO-DP (dynamic programming)
[256]	Optimal design for fast EV charging stations with wind, PV power, and energy storage system in the operating cycle of a week, month, or year	Combining MOPSO with other searching algorithms such as harmony search and cuckoo search
[257]	More complex non-linear and binary energy management problems	Constraint PSO-based MPC (model predictive control)
[130]	Integration of battery storage system to overcome the uncertainty of RERs in microgrid	Hybrid PSO optimization techniques

TABLE 5. (Continued.) Opportunities for new application under DSM.

[258]	Integration of different RESs and storage options to enhance the reliability of the power system	Not specified
[259]	 The scheduling of DSM through the coordination among different sectors in the presence of power grid, RE, ESS, and EV by embedding sensors and Internet of Things (IoT) modules in each participant Fog and cloud-based DSM employ various DR programs to achieve the desired balance between demand and supply Engagement of advanced, intelligent, and loads that have time as well as power flexibility for efficient energy management 	Hybrid bacterial foraging and particle swarm optimization (HBFPSO)
[260]	Increasing the number of customers, service providers, and industrial consumers considering the effect of integrating RES on the multi-stage incentive-based demand response program	Stackelberg-PSO
[261]	 Optimal powertrain sizing considering plug-in charging with SoC depleting mode (plug-in HEV) or zero urban emission zone running with electric-only operation in the urban part of the vehicle energy consumption calculation tool interurban cycle The effect of changing component size on vehicle mass and the corresponding variation in vehicle tractive resistance, power consumption, and optimally sized powertrain solution 	Not specified

lower PAR. Meanwhile, authors in [240] introduced interval number optimization for constraint violation based on tolerance degree to tackle minor uncertainty in residential load scheduling related to human behavior and weather conditions by combining PSO with an integer linear programming method. Considering the limitation of transforming the uncertain into a certain optimization, an extra function or information, such as probability distribution in stochastic programming, must be incorporated into household load scheduling, resulting in difficulty of optimization and demand for a large quantity of historical data. In specific scenarios, collecting data for a new home or in the absence of measurement instruments is challenging [240]. Furthermore, uncertainties about some input data, such as active and reactive load demands, as well as the unpredictable behavior of EV owners create unparalleled reliability and security issues to the overall distribution network. Therefore, probabilistic studies with a high degree of precision and tractable algorithms are necessary for the evaluation of uncertain behavior toward output variables of power system safety and balance operation [241].

VI. OPPORTUNITIES FOR NEW APPLICATION UNDER DSM

The previous literature reviews the significant work which has been conducted in the energy optimization field for different categories of DSM. Nevertheless, there are new applications that need to be explored as suggested in recent publications, so that the PSO-based method can be more relevant and developed in the real-life application of DSM. Thus, it can attract more consumers to participate and engage with the programs offered. Table 5 outlined new applications under DSM for opportunities to further research.

VII. CONCLUSION AND RECOMMENDATIONS

The effective implementation of DSM can encourage the production of low-cost, high-quality, and excellent services to consumers in addition to promoting supply-demand balance. This review paper presents a comprehensive overview of DSM activities in demand response and energy efficiency specifically. From the thorough review, DSM activities have facilitated in solving many of the challenges such as high electricity bills during peak hours, load curtailment in industrial buildings, and low efficiency of the system when integrated with RERs and diverse energy systems. The most commonly used DR program is TOU and RTP pricing to optimize electricity consumption and reduce peak demand. Since the cost of implementing a DR program is less expensive compared to increasing the capabilities of power generation to obtain the peak clipping, a DR program can be adopted to manage the demand and give benefits to both consumers and utilities. Meanwhile, EE measures allow consumers to save energy more efficiently.

To achieve the full potential of DSM activities optimally, the PSO-based method has been reviewed for optimization in various applications and solved the complexity of the models. After many years of development, most of the researchers developed the variants of PSO with the purpose to increase diversity, avoiding premature and enhancing the local search ability, thus improving the quality and robustness of the PSO algorithm. Observations from the study also indicated that improved and hybrid PSO-based models are computationally faster and resulted in better efficiency. The main optimization objectives including minimization of electricity bills, carbon emission, cost operation, peak demand, and discomfort of the occupant have been studied in most of the research works involving different types of consumers considering practical constraints. The proposed PSO-based algorithms proved to be effective in producing substantial cost savings while reducing the peak demand with the integration of the DR program and EE measures. Moreover, the key challenges of implementing PSO in DSM fields were discussed in terms of complexity and uncertainty in optimization, and several suggestions for new applications were identified to give insights for future research. The variants of PSO are used to address the high complexity and uncertainties of the DSM modelling within the acceptable timeframe while considering user comfort.

The application of AI algorithms is very important for the development of various fields in DSM. Therefore, more research on optimization models is necessary, especially when dealing with massive data. The integration of AI with data mining, IoT computing and blockchain, and advanced digital technologies must be combined to enhance the optimization models' performance instead of using them separately. The suggested research areas under DSM to review include the performance of intelligent buildings, smart energy management systems, and the application of demand response in electric power systems. With the increasing intermittent of RERs, the optimization problem of DSM can be more challenging considering the multi-objective function, complex constraints, uncertainties, and many variables. Thus, the enhancement of MA through different strategies such as hybridization, adopting the cooperative approach, and finding optimal parameter settings to deal with the complex problem can be further reviewed as the extension of this study. These applications can be discussed in major classifications of MA, such as evolutionary algorithms, human-based algorithms, physics algorithms, and swarm intelligence. The detailed review can outline the optimal configuration of the MA in solving the high complexity of the DSM optimization problem.

ACKNOWLEDGMENT

The authors would like to thank Universiti Teknikal Malaysia Melaka (UTeM) for all the support given.

REFERENCES

- R. A. Begum, A. Raihan, and M. N. M. Said, "Dynamic impacts of economic growth and forested area on carbon dioxide emissions in Malaysia," *Sustainability*, vol. 12, no. 22, pp. 1–15, 2020, doi: 10.3390/su12229375.
- [2] IPCC, 2022: Summary for Policymakers, "Climate change 2022: Mitigation of climate change," in *Contribution of Working Group III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change*, P. R. Shukla, J. Skea, R. Slade, A. Al Khourdajie, R. van Diemen, D. McCollum, M. Pathak, S. Some, P. Vyas, R. Fradera, M. Belkacemi, A. Hasija, G. Lisboa, S. Luz, and J. Malley, Eds. Cambridge, U.K.: Cambridge Univ. Press, 2022.
- [3] V. Masson-Delmotte, P. Zhai, H.-O. Pörtner, D. Roberts, J. Skea, P. R. Shukla, A. Pirani, W. Moufouma-Okia, C. Péan, R. Pidcock, S. Connors, and J. B. R. Matthews, "5°C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty," in *Global Warming of 1.5.C. An IPCC Special Report on the Impacts of Global Warming of 1*, Y. Chen, X. Zhou, M. Gomis, E. Lonnoy, T. Maycock, M. Tignor, T. Waterfield, Eds. Cambridge, U.K.: Cambridge Univ. Press, 2018, pp. 3–24.

- [4] L. Glowka, "Second conference of the parties," *Environ. Policy Law*, vol. 26, nos. 2–3, pp. 71–75, 2016.
- [5] V. Masson-Delmotte, "Global warming 1.5°C," Physical Science Basis, IPCC, Geneva, Switzerland, Working Group 1, 2018.
- [6] K. Levin, T. Fransen, C. Schumer, C. Davis, and S. Boehm. (2019). What Does 'Net-Zero Emissions' Mean? 8 Common Questions, Answered Explainer. [Online]. Available: https://www.wri.org/insights/net-zeroghg-emissions-questions-answered
- [7] S. Sorrell, "Reducing energy demand: A review of issues, challenges and approaches," *Renew. Sustain. Energy Rev.*, vol. 47, pp. 74–82, Jul. 2015, doi: 10.1016/j.rser.2015.03.002.
- [8] F. F. Said, K. A. Babatunde, M. A. Mahmoud, and R. A. Begum, "Decarbonizing the global electricity sector through demand-side management: A systematic critical review of policy responses," *J. Ekonomi Malaysia*, vol. 56, no. 1, pp. 71–91, 2022, doi: 10.17576/jem-2022-5601-06.
- [9] J. Ebrahimi and M. Abedini, "A two-stage framework for demand-side management and energy savings of various buildings in multi smart grid using robust optimization algorithms," *J. Building Eng.*, vol. 53, Aug. 2022, Art. no. 104486, doi: 10.1016/j.jobe.2022.104486.
- [10] A. Miara, C. Tarr, R. Spellman, C. J. Vörösmarty, and J. E. Macknick, "The power of efficiency: Optimizing environmental and social benefits through demand-side-management," *Energy*, vol. 76, pp. 502–512, Nov. 2014, doi: 10.1016/j.energy.2014.08.047.
- [11] D. I. Al Huneidi, F. Tahir, and S. G. Al-Ghamdi, "Energy modeling and photovoltaics integration as a mitigation measure for climate change impacts on energy demand," *Energy Rep.*, vol. 8, pp. 166–171, Jun. 2022, doi: 10.1016/j.egyr.2022.01.105.
- [12] P. Rafaj and S. Kypreos, "Internalisation of external cost in the power generation sector: Analysis with global multi-regional MARKAL model," *Energy Policy*, vol. 35, no. 2, pp. 828–843, Feb. 2007, doi: 10.1016/j.enpol.2006.03.003.
- [13] Q. Zhang and I. E. Grossmann, "Enterprise-wide optimization for industrial demand side management: Fundamentals, advances, and perspectives," *Chem. Eng. Res. Design*, vol. 116, pp. 114–131, Dec. 2016, doi: 10.1016/j.cherd.2016.10.006.
- [14] N. I. Zulkafli, M. A. Jelas, M. F. Bin Sulaima, M. F. Bin Sukri, M. M. T. Bin, and D. Hanak, A Demand Response Strategy for Air Compressors Network With Optimal Production and Energy Utilisation. Cambridge, U.K.: Cambridge Open Engage, 2022.
- [15] A. F. Meyabadi and M. H. Deihimi, "A review of demand-side management: Reconsidering theoretical framework," *Renew. Sustain. Energy Rev.*, vol. 80, pp. 367–379, Dec. 2017, doi: 10.1016/j.rser.2017.05.207.
- [16] X. Yan, Y. Ozturk, Z. Hu, and Y. Song, "A review on price-driven residential demand response," *Renew. Sustain. Energy Rev.*, vol. 96, pp. 411–419, Nov. 2018, doi: 10.1016/j.rser.2018.08.003.
- [17] C. A. Cardoso, J. Torriti, and M. Lorincz, "Making demand side response happen: A review of barriers in commercial and public organisations," *Energy Res. Social Sci.*, vol. 64, Jun. 2020, Art. no. 101443, doi: 10.1016/j.erss.2020.101443.
- [18] C. Yang, C. Meng, and K. Zhou, "Residential electricity pricing in China: The context of price-based demand response," *Renew. Sustain. Energy Rev.*, vol. 81, pp. 2870–2878, Jan. 2018, doi: 10.1016/j.rser.2017.06.093.
- [19] D. Kanakadhurga and N. Prabaharan, "Demand side management in microgrid: A critical review of key issues and recent trends," *Renew. Sustain. Energy Rev.*, vol. 156, Mar. 2022, Art. no. 111915, doi: 10.1016/j.rser.2021.111915.
- [20] L. Tronchin, M. Manfren, and B. Nastasi, "Energy efficiency, demand side management and energy storage technologies—A critical analysis of possible paths of integration in the built environment," *Renew. Sustain. Energy Rev.*, vol. 95, pp. 341–353, Nov. 2018, doi: 10.1016/j.rser.2018.06.060.
- [21] S. Chatterjee, N. Rafa, and A. Nandy, "Welfare, development, and costefficiency: A global synthesis on incentivizing energy efficiency measures through co-benefits," *Energy Res. Social Sci.*, vol. 89, Jul. 2022, Art. no. 102666, doi: 10.1016/j.erss.2022.102666.
- [22] J. Torriti, M. G. Hassan, and M. Leach, "Demand response experience in Europe: Policies, programmes and implementation," *Energy*, vol. 35, no. 4, pp. 1575–1583, Apr. 2010, doi: 10.1016/j.energy.2009.05.021.
- [23] R. Carmichael, R. Gross, R. Hanna, A. Rhodes, and T. Green, "The demand response technology cluster: Accelerating U.K. residential consumer engagement with time-of-use tariffs, electric vehicles and smart meters via digital comparison tools," *Renew. Sustain. Energy Rev.*, vol. 139, Apr. 2021, Art. no. 110701, doi: 10.1016/j.rser.2020.110701.

- [24] E. Sarker, M. Seyedmahmoudian, E. Jamei, B. Horan, and A. Stojcevski, "Optimal management of home loads with renewable energy integration and demand response strategy," *Energy*, vol. 210, Nov. 2020, Art. no. 118602, doi: 10.1016/j.energy.2020.118602.
- [25] A. Lyden, R. Pepper, and P. G. Tuohy, "A modelling tool selection process for planning of community scale energy systems including storage and demand side management," *Sustain. Cities Soc.*, vol. 39, pp. 674–688, May 2018, doi: 10.1016/j.scs.2018.02.003.
- [26] S. Rinaldi, M. Pasetti, E. Sisinni, F. Bonafini, P. Ferrari, M. Rizzi, and A. Flammini, "On the mobile communication requirements for the demand-side management of electric vehicles," *Energies*, vol. 11, no. 5, p. 1220, May 2018, doi: 10.3390/en11051220.
- [27] D. Mariano-Hernández, L. Hernández-Callejo, A. Zorita-Lamadrid, O. Duque-Pérez, and F. Santos García, "A review of strategies for building energy management system: Model predictive control, demand side management, optimization, and fault detect & diagnosis," *J. Building Eng.*, vol. 33, Jan. 2021, Art. no. 101692, doi: 10.1016/j.jobe.2020.101692.
- [28] R. Liaqat, I. A. Sajjad, M. Waseem, and H. H. Alhelou, "Appliance level energy characterization of residential electricity demand: Prospects, challenges and recommendations," *IEEE Access*, vol. 9, pp. 148676–148697, 2021, doi: 10.1109/ACCESS.2021.3123196.
- [29] I. Antonopoulos, V. Robu, B. Couraud, D. Kirli, S. Norbu, A. Kiprakis, D. Flynn, S. Elizondo-Gonzalez, and S. Wattam, "Artificial intelligence and machine learning approaches to energy demand-side response: A systematic review," *Renew. Sustain. Energy Rev.*, vol. 130, Sep. 2020, Art. no. 109899, doi: 10.1016/j.rser.2020.109899.
- [30] R. Vinuesa, H. Azizpour, I. Leite, M. Balaam, V. Dignum, S. Domisch, A. Felländer, S. D. Langhans, M. Tegmark, and F. Fuso Nerini, "The role of artificial intelligence in achieving the sustainable development goals," *Nature Commun.*, vol. 11, no. 1, p. 233, Jan. 2020, doi: 10.1038/s41467-019-14108-y.
- [31] Y. He, N. Liao, and K. Lin, "Can China's industrial sector achieve energy conservation and emission reduction goals dominated by energy efficiency enhancement? A multi-objective optimization approach," *Energy Policy*, vol. 149, Feb. 2021, Art. no. 112108, doi: 10.1016/j.enpol.2020.112108.
- [32] B. Chreim, M. Esseghir, and L. Merghem-Boulahia, "LOSISH—Load scheduling in smart homes based on demand response: Application to smart grids," *Appl. Energy*, vol. 323, Oct. 2022, Art. no. 119606, doi: 10.1016/j.apenergy.2022.119606.
- [33] A. Shewale, A. Mokhade, N. Funde, and N. D. Bokde, "An overview of demand response in smart grid and optimization techniques for efficient residential appliance scheduling problem," *Energies*, vol. 13, no. 16, p. 4266, Aug. 2020, doi: 10.3390/en13164266.
- [34] R. Faia, P. Faria, Z. Vale, and J. Spinola, "Demand response optimization using particle swarm algorithm considering optimum battery energy storage schedule in a residential house," *Energies*, vol. 12, no. 9, p. 1645, Apr. 2019, doi: 10.3390/en12091645.
- [35] W. Y. Wang, Z. H. Xu, Y. H. Fan, D. D. Pan, P. Lin, and X. T. Wang, "Disturbance inspired equilibrium optimizer with application to constrained engineering design problems," *Appl. Math. Model.*, vol. 116, pp. 254–276, Apr. 2023, doi: 10.1016/j.apm.2022.11.016.
- [36] Y. Shen, C. Zhang, F. S. Gharehchopogh, and S. Mirjalili, "An improved whale optimization algorithm based on multi-population evolution for global optimization and engineering design problems," *Exp. Syst. Appl.*, vol. 215, Apr. 2023, Art. no. 119269, doi: 10.1016/j.eswa.2022.119269.
- [37] O. E. Turgut and M. S. Turgut, "Local search enhanced Aquila optimization algorithm ameliorated with an ensemble of wavelet mutation strategies for complex optimization problems," *Math. Comput. Simul.*, vol. 206, pp. 302–374, Apr. 2023, doi: 10.1016/j.matcom.2022.11.020.
- [38] M. Nutakki and S. Mandava, "Review on optimization techniques and role of artificial intelligence in home energy management systems," *Eng. Appl. Artif. Intell.*, vol. 119, Mar. 2023, Art. no. 105721, doi: 10.1016/j.engappai.2022.105721.
- [39] O. W. Khalid, N. A. M. Isa, and H. A. M. Sakim, "Emperor penguin optimizer: A comprehensive review based on state-of-the-art meta-heuristic algorithms," *Alexandria Eng. J.*, vol. 63, pp. 487–526, Jan. 2023, doi: 10.1016/j.aej.2022.08.013.
- [40] M. F. Sulaima, N. Y. Dahlan, W. N. A. W. Hanapi, and M. M. N. Din, "Malaysia residential load profile management based on time of use tariff using ant colony optimization algorithm," *J. Sustainability Sci. Manage.*, vol. 17, no. 3, pp. 245–255, 2022.

- [41] M. F. Sulaima, N. Baharin, A. F. A. Kadir, N. Salim, and E. E. Hassan, "Investigation of electricity load shifting under various tariff design using ant colony optimization algorithm," *Indonesian J. Electr. Eng. Comput. Sci.*, vol. 28, no. 1, pp. 1–11, 2022, doi: 10.11591/ijeecs.v28.i1. pp1-11.
- [42] A. Niccolai, G. G. Taje, D. Mosca, F. Trombello, and E. Ogliari, "Industrial demand-side management by means of differential evolution considering energy price and labour cost," *Mathematics*, vol. 10, no. 19, pp. 1–16, 2022, doi: 10.3390/math10193605.
- [43] W. Chiu, J. Hsieh, and C. Chen, "Pareto optimal demand response based on energy costs and load factor in smart grid," *IEEE Trans. Ind. Informat.*, vol. 16, no. 3, pp. 1811–1822, Mar. 2020, doi: 10.1109/TII.2019.2928520.
- [44] A. M. Jasim, B. H. Jasim, S. Mohseni, and A. C. Brent, "Consensusbased dispatch optimization of a microgrid considering metaheuristic-based demand response scheduling and network packet loss characterization," *Energy AI*, vol. 11, Jan. 2023, Art. no. 100212, doi: 10.1016/j.egyai.2022.100212.
- [45] X. Wang, W. Song, H. Wu, H. Liang, and A. Saboor, "Microgrid operation relying on economic problems considering renewable sources, storage system, and demand-side management using developed gray wolf optimization algorithm," *Energy*, vol. 248, Jun. 2022, Art. no. 123472, doi: 10.1016/j.energy.2022.123472.
- [46] A. Agrawal, S. N. Pandey, L. Srivastava, P. Walde, R. K. Saket, and B. Khan, "Multiobjective salp swarm algorithm approach for transmission congestion management," *Int. Trans. Electr. Energy Syst.*, vol. 2022, pp. 1–17, Dec. 2022, doi: 10.1155/2022/8256908.
- [47] I. Hussain, M. Ullah, I. Ullah, A. Bibi, M. Naeem, M. Singh, and D. Singh, "Optimizing energy consumption in the home energy management system via a bio-inspired dragonfly algorithm and the genetic algorithm," *Electronics*, vol. 9, no. 3, p. 406, Feb. 2020, doi: 10.3390/electronics9030406.
- [48] B. N. Silva and K. Han, "Mutation operator integrated ant colony optimization based domestic appliance scheduling for lucrative demand side management," *Future Gener. Comput. Syst.*, vol. 100, pp. 557–568, Nov. 2019, doi: 10.1016/j.future.2019.05.052.
- [49] M. Irfan, C. A. Ul Hassan, F. Althobiani, N. Ayub, R. J. U. H. Khan, E. I. Ghandourah, M. A. Almas, S. M. Ghonaim, V. R. Shamji, and S. Rahman, "Power scheduling with max user comfort in smart home: Performance analysis and tradeoffs," *Comput. Syst. Sci. Eng.*, vol. 46, no. 2, pp. 1723–1740, 2023, doi: 10.32604/csse.2023.035141.
- [50] Z. A. Khan, A. Khalid, N. Javaid, A. Haseeb, T. Saba, and M. Shafiq, "Exploiting nature-inspired-based artificial intelligence techniques for coordinated day-ahead scheduling to efficiently manage energy in smart grid," *IEEE Access*, vol. 7, pp. 140102–140125, 2019, doi: 10.1109/ACCESS.2019.2942813.
- [51] M. M. Rosso, R. Cucuzza, A. Aloisio, and G. C. Marano, "Enhanced multi-strategy particle swarm optimization for constrained problems with an evolutionary-strategies-based unfeasible local search operator," *Appl. Sci.*, vol. 12, no. 5, p. 2285, 2022, doi: 10.3390/app12052285.
- [52] J. Aguilar, A. Garces-Jimenez, M. D. R-Moreno, and R. García, "A systematic literature review on the use of artificial intelligence in energy selfmanagement in smart buildings," *Renew. Sustain. Energy Rev.*, vol. 151, Nov. 2021, Art. no. 111530, doi: 10.1016/j.rser.2021.111530.
- [53] H. Farzaneh, L. Malehmirchegini, A. Bejan, T. Afolabi, A. Mulumba, and P. P. Daka, "Artificial intelligence evolution in smart buildings for energy efficiency," *Appl. Sci.*, vol. 11, no. 2, p. 763, 2021, doi: 10.3390/app11020763.
- [54] J. R. Vázquez-Canteli and Z. Nagy, "Reinforcement learning for demand response: A review of algorithms and modeling techniques," *Appl. Energy*, vol. 235, pp. 1072–1089, Feb. 2019, doi: 10.1016/j.apenergy.2018.11.002.
- [55] S. Panda, S. Mohanty, P. K. Rout, B. K. Sahu, M. Bajaj, H. M. Zawbaa, and S. Kamel, "Residential demand side management model, optimization and future perspective: A review," *Energy Rep.*, vol. 8, pp. 3727–3766, Nov. 2022, doi: 10.1016/j.egyr.2022.02.300.
- [56] M. Sameti and F. Haghighat, "Optimization approaches in district heating and cooling thermal network," *Energy Buildings*, vol. 140, pp. 121–130, Apr. 2017, doi: 10.1016/j.enbuild.2017.01.062.
- [57] E. Guelpa and V. Verda, "Demand response and other demand side management techniques for district heating: A review," *Energy*, vol. 219, Mar. 2021, Art. no. 119440, doi: 10.1016/j.energy.2020.119440.

- [58] G. S. Thirunavukkarasu, M. Seyedmahmoudian, E. Jamei, B. Horan, S. Mekhilef, and A. Stojcevski, "Role of optimization techniques in microgrid energy management systems—A review," *Energy Strategy Rev.*, vol. 43, Sep. 2022, Art. no. 100899, doi: 10.1016/j.esr.2022.100899.
- [59] C. Menos-Aikateriniadis, I. Lamprinos, and P. S. Georgilakis, "Particle swarm optimization in residential demand-side management: A review on scheduling and control algorithms for demand response provision," *Energies*, vol. 15, no. 6, p. 2211, Mar. 2022, doi: 10.3390/en15062211.
- [60] S. Alam, G. Dobbie, Y. S. Koh, P. Riddle, and S. Ur Rehman, "Research on particle swarm optimization based clustering: A systematic review of literature and techniques," *Swarm Evol. Comput.*, vol. 17, pp. 1–13, Aug. 2014, doi: 10.1016/j.swevo.2014.02.001.
- [61] J. S. Russell and P. Norvig, Artificial Intelligence: A Modern Approach (Artificial Intelligence), 3rd, Ed., Upper Saddle River, NJ, USA: Prentice-Hall, 2010.
- [62] A. Darwish, "Bio-inspired computing: Algorithms review, deep analysis, and the scope of applications," *Future Comput. Informat. J.*, vol. 3, no. 2, pp. 231–246, Dec. 2018, doi: 10.1016/j.fcij.2018.06.001.
- [63] Z.-G. Liu, X.-H. Ji, Y. Yang, and H.-T. Cheng, "Multi-technique diversity-based particle-swarm optimization," *Inf. Sci.*, vol. 577, pp. 298–323, Oct. 2021, doi: 10.1016/j.ins.2021.07.006.
- [64] D. Tian and Z. Shi, "MPSO: Modified particle swarm optimization and its applications," *Swarm Evol. Comput.*, vol. 41, pp. 49–68, Aug. 2018, doi: 10.1016/j.swevo.2018.01.011.
- [65] H. Zhang, Q. Cao, H. Gao, P. Wang, W. Zhang, and N. Yousefi, "Optimum design of a multi-form energy hub by applying particle swarm optimization," *J. Cleaner Prod.*, vol. 260, Jul. 2020, Art. no. 121079, doi: 10.1016/j.jclepro.2020.121079.
- [66] Z. Ceylan, "Short-term prediction of COVID-19 spread using grey rolling model optimized by particle swarm optimization," *Appl. Soft Comput.*, vol. 109, Sep. 2021, Art. no. 107592, doi: 10.1016/j.asoc.2021.107592.
- [67] G. Singh and A. Singh, "Extension of particle swarm optimization algorithm for solving transportation problem in fuzzy environment," *Appl. Soft Comput.*, vol. 110, Oct. 2021, Art. no. 107619, doi: 10.1016/j.asoc.2021.107619.
- [68] C. P. Mohanty, S. S. Mahapatra, and M. R. Singh, "A particle swarm approach for multi-objective optimization of electrical discharge machining process," *J. Intell. Manuf.*, vol. 27, no. 6, pp. 1171–1190, Dec. 2016, doi: 10.1007/s10845-014-0942-3.
- [69] I. Mazhoud, K. Hadj-Hamou, J. Bigeon, and P. Joyeux, "Particle swarm optimization for solving engineering problems: A new constraint-handling mechanism," *Eng. Appl. Artif. Intell.*, vol. 26, no. 4, pp. 1263–1273, Apr. 2013, doi: 10.1016/j.engappai.2013.02.002.
- [70] N. Kinhekar, N. P. Padhy, and H. O. Gupta, "Particle swarm optimization based demand response for residential consumers," in *Proc. IEEE Power Energy Soc. Gen. Meeting*, Jul. 2015, pp. 1–5, doi: 10.1109/PESGM.2015.7286466.
- [71] Y. del Valle, G. K. Venayagamoorthy, S. Mohagheghi, J.-C. Hernandez, and R. G. Harley, "Particle swarm optimization: Basic concepts, variants and applications in power systems," *IEEE Trans. Evol. Comput.*, vol. 12, no. 2, pp. 171–195, Apr. 2008.
- [72] D. Wang, D. Tan, and L. Liu, "Particle swarm optimization algorithm: An overview," *Soft Comput.*, vol. 22, no. 2, pp. 387–408, Jan. 2018, doi: 10.1007/s00500-016-2474-6.
- [73] F. Wang, H. Zhang, K. Li, Z. Lin, J. Yang, and X.-L. Shen, "A hybrid particle swarm optimization algorithm using adaptive learning strategy," *Inf. Sci.*, vols. 436–437, pp. 162–177, Apr. 2018, doi: 10.1016/j.ins.2018.01.027.
- [74] E. H. Houssein, A. G. Gad, K. Hussain, and P. N. Suganthan, "Major advances in particle swarm optimization: Theory, analysis, and application," *Swarm Evol. Comput.*, vol. 63, Jun. 2021, Art. no. 100868, doi: 10.1016/j.swevo.2021.100868.
- [75] J. Gou, Y.-X. Lei, W.-P. Guo, C. Wang, Y.-Q. Cai, and W. Luo, "A novel improved particle swarm optimization algorithm based on individual difference evolution," *Appl. Soft Comput.*, vol. 57, pp. 468–481, Aug. 2017, doi: 10.1016/j.asoc.2017.04.025.
- [76] Z. Liu and T. Nishi, "Strategy dynamics particle swarm optimizer," *Inf. Sci.*, vol. 582, pp. 665–703, Jan. 2022, doi: 10.1016/j.ins.2021. 10.028.
- [77] X. Zhang, X. Wang, Q. Kang, and J. Cheng, "Differential mutation and novel social learning particle swarm optimization algorithm," *Inf. Sci.*, vol. 480, pp. 109–129, Apr. 2019, doi: 10.1016/j.ins.2018.12.030.

- [78] J. J. Liang, A. K. Qin, P. N. Suganthan, and S. Baskar, "Comprehensive learning particle swarm optimizer for global optimization of multimodal functions," *IEEE Trans. Evol. Comput.*, vol. 10, no. 3, pp. 281–295, Jun. 2006, doi: 10.1109/TEVC.2005.857610.
- [79] A. Zhang, H. Xu, W. Bi, and S. Xu, "Adaptive mutant particle swarm optimization based precise cargo airdrop of unmanned aerial vehicles," *Appl. Soft Comput.*, vol. 130, Nov. 2022, Art. no. 109657, doi: 10.1016/j.asoc.2022.109657.
- [80] R. Pinto, R. J. Bessa, and M. A. Matos, "Multi-period flexibility forecast for low voltage prosumers," *Energy*, vol. 141, pp. 2251–2263, Dec. 2017, doi: 10.1016/j.energy.2017.11.142.
- [81] F. van den Bergh and A. P. Engelbrecht, "A cooperative approach to particle swarm optimization," *IEEE Trans. Evol. Comput.*, vol. 8, no. 3, pp. 225–239, Jun. 2004, doi: 10.1109/TEVC.2004.826069.
- [82] W. Fang, Q. Zhang, H. Lu, and J. C.-W. Lin, "High-utility itemsets mining based on binary particle swarm optimization with multiple adjustment strategies," *Appl. Soft Comput.*, vol. 124, Jul. 2022, Art. no. 109073, doi: 10.1016/j.asoc.2022.109073.
- [83] M. Jamalipour, M. Gharib, R. Sayareh, and F. Khoshahval, "PWR power distribution flattening using quantum particle swarm intelligence," *Ann. Nucl. Energy*, vol. 56, pp. 143–150, Jun. 2013, doi: 10.1016/j.anucene.2013.01.026.
- [84] A. R. Jordehi, "Binary particle swarm optimisation with quadratic transfer function: A new binary optimisation algorithm for optimal scheduling of appliances in smart homes," *Appl. Soft Comput.*, vol. 78, pp. 465–480, May 2019, doi: 10.1016/j.asoc.2019.03.002.
- [85] D. C. Valencia-Rodríguez and C. A. Coello Coello, "Influence of the number of connections between particles in the performance of a multiobjective particle swarm optimizer," *Swarm Evol. Comput.*, vol. 77, Mar. 2023, Art. no. 101231, doi: 10.1016/j.swevo.2023.101231.
- [86] F. Javidrad and M. Nazari, "A new hybrid particle swarm and simulated annealing stochastic optimization method," *Appl. Soft Comput.*, vol. 60, pp. 634–654, Nov. 2017, doi: 10.1016/j.asoc.2017.07.023.
- [87] Z. Wang, J. Tian, and K. Feng, "Optimal allocation of regional water resources based on simulated annealing particle swarm optimization algorithm," *Energy Rep.*, vol. 8, pp. 9119–9126, Nov. 2022, doi: 10.1016/j.egyr.2022.07.033.
- [88] A. Rezaee Jordehi, "Enhanced leader particle swarm optimisation (ELPSO): An efficient algorithm for parameter estimation of photovoltaic (PV) cells and modules," *Sol. Energy*, vol. 159, pp. 78–87, Jan. 2018, doi: 10.1016/j.solener.2017.10.063.
- [89] P. Siano, "Demand response and smart grids—A survey," *Renew. Sustain. Energy Rev.*, vol. 30, pp. 461–478, Feb. 2014, doi: 10.1016/j.rser.2013.10.022.
- [90] S. Abdollahi Kakroudi, R. Ebrahimi, and A. Ahmadi, "Real-time undervoltage load shedding method considering integer-value load model and feeder participation in demand response," *Electric Power Syst. Res.*, vol. 217, Apr. 2023, Art. no. 109115, doi: 10.1016/j.epsr.2023.109115.
- [91] S. A. Mansouri, E. Nematbakhsh, A. Ahmarinejad, A. R. Jordehi, M. S. Javadi, and S. A. A. Matin, "A multi-objective dynamic framework for design of energy hub by considering energy storage system, power-to-gas technology and integrated demand response program," *J. Energy Storage*, vol. 50, Jun. 2022, Art. no. 104206, doi: 10.1016/j.est.2022.104206.
- [92] J. S. Vardakas, N. Zorba, and C. V. Verikoukis, "A survey on demand response programs in smart grids: Pricing methods and optimization algorithms," *IEEE Commun. Surveys Tuts.*, vol. 17, no. 1, pp. 152–178, 1st Quart., 2015, doi: 10.1109/COMST.2014.2341586.
- [93] L. Wang, J. Lin, H. Dong, Y. Wang, and M. Zeng, "Demand response comprehensive incentive mechanism-based multi-time scale optimization scheduling for park integrated energy system," *Energy*, vol. 270, May 2023, Art. no. 126893, doi: 10.1016/j.energy.2023.126893.
- [94] A. R. Jordehi, "Optimisation of demand response in electric power systems, a review," *Renew. Sustain. Energy Rev.*, vol. 103, pp. 308–319, Apr. 2019, doi: 10.1016/j.rser.2018.12.054.
- [95] N. Venkatesan, J. Solanki, and S. K. Solanki, "Residential demand response model and impact on voltage profile and losses of an electric distribution network," *Appl. Energy*, vol. 96, pp. 84–91, Aug. 2012, doi: 10.1016/j.apenergy.2011.12.076.
- [96] D. Darwazeh, J. Duquette, B. Gunay, I. Wilton, and S. Shillinglaw, "Review of peak load management strategies in commercial buildings," *Sustain. Cities Soc.*, vol. 77, Feb. 2022, Art. no. 103493, doi: 10.1016/j.scs.2021.103493.

- [97] F. Guang, Y. He, and L. Wen, "Impacts of hybrid time-varying tariffs on residential electricity demand: The case of Zhejiang province," *Utilities Policy*, vol. 61, Dec. 2019, Art. no. 100969, doi: 10.1016/j.jup.2019.100969.
- [98] M. P. Abdullah, N. S. M. Nazar, M. Y. Hassan, and F. Hussin, "Optimizing time of use (ToU) electricity pricing in regulated market," *J. Teknologi*, vol. 78, nos. 5–7, pp. 49–54, 2016, doi: 10.11113/jt.v78.8712.
- [99] B. Parrish, P. Heptonstall, R. Gross, and B. K. Sovacool, "A systematic review of motivations, enablers and barriers for consumer engagement with residential demand response," *Energy Policy*, vol. 138, Mar. 2020, Art. no. 111221, doi: 10.1016/j.enpol.2019.111221.
- [100] M. F. Sulaima, N. Y. Dahlan, Z. M. Yasin, M. M. Rosli, Z. Omar, and M. Y. Hassan, "A review of electricity pricing in peninsular malaysia: Empirical investigation about the appropriateness of enhanced time of use (ETOU) electricity tariff," *Renew. Sustain. Energy Rev.*, vol. 110, pp. 348–367, Aug. 2019, doi: 10.1016/j.rser.2019.04.075.
- [101] J. Torriti, "Price-based demand side management: Assessing the impacts of time-of-use tariffs on residential electricity demand and peak shifting in northern Italy," *Energy*, vol. 44, no. 1, pp. 576–583, Aug. 2012, doi: 10.1016/j.energy.2012.05.043.
- [102] Y. Zhang, M. M. Islam, Z. Sun, S. Yang, C. Dagli, and H. Xiong, "Optimal sizing and planning of onsite generation system for manufacturing in critical peaking pricing demand response program," *Int. J. Prod. Econ.*, vol. 206, pp. 261–267, Dec. 2018, doi: 10.1016/j.ijpe.2018. 10.011.
- [103] T. R. Olorunfemi and N. Nwulu, "A review of demand response techniques and operational limitations," in *Proc. Int. Conf. Comput. Techn., Electron. Mech. Syst. (CTEMS)*, Dec. 2018, pp. 442–445, doi: 10.1109/CTEMS.2018.8769181.
- [104] Y. Wang and L. Li, "Critical peak electricity pricing for sustainable manufacturing: Modeling and case studies," *Appl. Energy*, vol. 175, pp. 40–53, Aug. 2016, doi: 10.1016/j.apenergy.2016.04.100.
- [105] Y. Li, W. Gao, Y. Ruan, and Y. Ushifusa, "Demand response of customers in Kitakyushu smart community project to critical peak pricing of electricity," *Energy Buildings*, vol. 168, pp. 251–260, Jun. 2018, doi: 10.1016/j.enbuild.2018.03.029.
- [106] H. Y. Song, G. S. Lee, and Y. T. Yoon, "Optimal operation of critical peak pricing for an energy retailer considering balancing costs," *Energies*, vol. 12, no. 24, p. 4658, Dec. 2019, doi: 10.3390/en12244658.
- [107] L. Yi, P. Langlotz, M. Hussong, M. Glatt, F. J. P. Sousa, and J. C. Aurich, "An integrated energy management system using double deep Q-learning and energy storage equipment to reduce energy cost in manufacturing under real-time pricing condition: A case study of scale-model factory," *CIRP J. Manuf. Sci. Technol.*, vol. 38, pp. 844–860, Aug. 2022, doi: 10.1016/j.cirpj.2022.07.009.
- [108] I. Panapakidis, N. Asimopoulos, A. Dagoumas, and G. C. Christoforidis, "An improved fuzzy C-means algorithm for the implementation of demand side management measures," *Energies*, vol. 10, no. 9, p. 1407, Sep. 2017, doi: 10.3390/en10091407.
- [109] Z. Hu, J.-H. Kim, J. Wang, and J. Byrne, "Review of dynamic pricing programs in the U.S. and Europe: Status quo and policy recommendations," *Renew. Sustain. Energy Rev.*, vol. 42, pp. 743–751, Feb. 2015, doi: 10.1016/j.rser.2014.10.078.
- [110] S. Ambec and C. Crampes, "Real-time electricity pricing to balance green energy intermittency," *Energy Econ.*, vol. 94, Feb. 2021, Art. no. 105074, doi: 10.1016/j.eneco.2020.105074.
- [111] A. Anees and Y.-P.-P. Chen, "True real time pricing and combined power scheduling of electric appliances in residential energy management system," *Appl. Energy*, vol. 165, pp. 592–600, Mar. 2016, doi: 10.1016/j.apenergy.2015.12.103.
- [112] A. Leon-Garcia, "Optimal residential load control with price prediction in real-time electricity pricing environments," *IEEE Trans. Smart Grid*, vol. 1, no. 2, pp. 120–133, Sep. 2010. [Online]. Available: http:// ieeexplore.ieee.org/lpdocs/epic03/wrapper.htm?arnumber=5540263
- [113] J. Thakur and B. Chakraborty, "Impact of increased solar penetration on bill savings of net metered residential consumers in India," *Energy*, vol. 162, pp. 776–786, Nov. 2018, doi: 10.1016/j.energy.2018. 08.025.
- [114] M. S. Javadi, M. Gough, M. Lotfi, A. E. Nezhad, S. F. Santos, and J. P. S. Catalão, "Optimal self-scheduling of home energy management system in the presence of photovoltaic power generation and batteries," *Energy*, vol. 210, Nov. 2020, Art. no. 118568, doi: 10.1016/j.energy.2020.118568.

- [115] S. Yilmaz, C. Chanez, P. Cuony, and M. K. Patel, "Analysing utilitybased direct load control programmes for heat pumps and electric vehicles considering customer segmentation," *Energy Policy*, vol. 164, May 2022, Art. no. 112900, doi: 10.1016/j.enpol.2022.112900.
- [116] J. A. R. Curiel and J. Thakur, "A novel approach for direct load control of residential air conditioners for demand side management in developing regions," *Energy*, vol. 258, Nov. 2022, Art. no. 124763, doi: 10.1016/j.energy.2022.124763.
- [117] H. Dagdougui, A. Ouammi, and R. Sacile, "Towards a concept of cooperating power network for energy management and control of microgrids," in *Microgrid*, M. S. Mahmoud, Ed. Butterworth-Heinemann, 2017, ch. 8, pp. 231–262, doi: 10.1016/B978-0-08-101753-1.00008-5.
- [118] S. Y. Choi and K. Y. Lee, "Direct load control system design using active database," in *Proc. 19th World Congr. Int. Fed. Autom. Control Cape Town*, South Africa, Aug. 2014, pp. 9936–9941.
- [119] Benefits of Demand Response in Electricity Markets and Recommendations for Achieving Them, A Report to the United States Congress Pursuant to Section 1252 of the Energy Policy Act of 2005, U.S. Dept. Energy, Washington, DC, USA, Feb. 2006.
- [120] H. A. Aalami, M. P. Moghaddam, and G. R. Yousefi, "Demand response modeling considering interruptible/curtailable loads and capacity market programs," *Appl. Energy*, vol. 87, no. 1, pp. 243–250, Jan. 2010, doi: 10.1016/j.apenergy.2009.05.041.
- [121] A. Ng'uni and L. A. Tuan, "Interruptible load and demand response: Worldwide picture and the situation in Sweden," in *Proc. 38th North Amer. Power Symp.*, Carbondale, IL, USA, 2006, pp. 121–127, doi: 10.1109/NAPS.2006.360133.
- [122] H. A. Aalami and A. Khatibzadeh, "Regulation of market clearing price based on nonlinear models of demand bidding and emergency demand response programs," *Int. Trans. Electr. Energy Syst.*, vol. 20, no. 11, pp. 1–6, 2016, doi: 10.1002/etep.2216.
- [123] M. Fahrioglu and F. L. Alvarado, "Designing incentive compatible contracts for effective demand management," *IEEE Trans. Power Syst.*, vol. 15, no. 4, pp. 1255–1260, 2000, doi: 10.1109/59.898098.
- [124] H. Aalami, G. R. Yousefi, and M. P. Moghadam, "Demand response model considering EDRP and TOU programs," in *Proc. IEEE/PES Transmiss. Distrib. Conf. Expo.*, Apr. 2008, pp. 1–6, doi: 10.1109/TDC.2008.4517059.
- [125] Y.-K. Wu and K.-T. Tang, "Frequency support by demand response– review and analysis," *Energy Procedia*, vol. 156, pp. 327–331, Jan. 2019, doi: 10.1016/j.egypro.2018.11.150.
- [126] The Power to Choose: Demand Response in Liberalised Electricity Markets, International Energy Agency, France, 2003.
- [127] R. Alasseri, T. J. Rao, and K. J. Sreekanth, "Institution of incentivebased demand response programs and prospective policy assessments for a subsidized electricity market," *Renew. Sustain. Energy Rev.*, vol. 117, Jan. 2020, Art. no. 109490, doi: 10.1016/j.rser.2019.109490.
- [128] J. Wang, S. Kennedy, and J. Kirtley, "A new wholesale bidding mechanism for enhanced demand response in smart grids," in *Proc. Innov. Smart Grid Technol. (ISGT)*, Jan. 2010, pp. 1–9, doi: 10.1109/ISGT.2010.5434766.
- [129] N. Bayat and J. Park, "Particle swarm optimization based demand response using artificial neural network based load prediction," in *Proc. North Amer. Power Symp. (NAPS)*, Salt Lake City, UT, USA, Oct. 2022, pp. 1–5, doi: 10.1109/naps56150.2022.10012263.
- [130] M. A. S. Hassan, "Dynamic price-based demand response through linear regression for microgrids with renewable energy resources," *Energies*, vol. 15, no. 4, pp. 1–17, 2022, doi: 10.3390/en15041385.
- [131] M. A. Khan, A. M. Saleh, M. Waseem, and I. A. Sajjad, "Artificial intelligence enabled demand response: Prospects and challenges in smart grid environment," *IEEE Access*, vol. 11, pp. 1477–1505, Dec. 2023, doi: 10.1109/ACCESS.2022.3231444.
- [132] A. Mohammad, M. Zuhaib, I. Ashraf, M. Alsultan, S. Ahmad, A. Sarwar, and M. Abdollahian, "Integration of electric vehicles and energy storage system in home energy management system with home to grid capability," *Energies*, vol. 14, no. 24, p. 8557, Dec. 2021, doi: 10.3390/en14248557.
- [133] D. Kanakadhurga and N. Prabaharan, "Demand response-based peer-topeer energy trading among the prosumers and consumers," *Energy Rep.*, vol. 7, pp. 7825–7834, Nov. 2021, doi: 10.1016/j.egyr.2021.09.074.
- [134] J. Ding, S. Schulz, L. Shen, U. Buscher, and Z. Lü, "Energy aware scheduling in flexible flow shops with hybrid particle swarm optimization," *Comput. Operations Res.*, vol. 125, Jan. 2021, Art. no. 105088, doi: 10.1016/j.cor.2020.105088.

- [135] D. Bahmanyar, N. Razmjooy, and S. Mirjalili, "Multi-objective scheduling of IoT-enabled smart homes for energy management based on arithmetic optimization algorithm: A node-RED and NodeMCU module-based technique," *Knowledge-Based Syst.*, vol. 247, Jul. 2022, Art. no. 108762, doi: 10.1016/j.knosys.2022.108762.
- [136] H. Molavi and M. M. Ardehali, "Utility demand response operation considering day-of-use tariff and optimal operation of thermal energy storage system for an industrial building based on particle swarm optimization algorithm," *Energy Buildings*, vol. 127, pp. 920–929, Sep. 2016, doi: 10.1016/j.enbuild.2016.06.056.
- [137] L. Priya and V. Gomathi, "Analysis of price and incentive based demand response programs on unit commitment using particle swarm optimization," *Int. J. Electr. Electron. Res.*, vol. 10, no. 2, pp. 353–359, Jun. 2022, doi: 10.37391/IJEER.100248.
- [138] Y. Wang and L. Li, "Manufacturing profit maximization under timevarying electricity and labor pricing," *Comput. Ind. Eng.*, vol. 104, pp. 23–34, Feb. 2017, doi: 10.1016/j.cie.2016.12.011.
- [139] K. Zhu, N. Lu, J. Zheng, G. Sun, and F. Mei, "Optimal day-ahead scheduling for commercial building-level consumers under TOU and demand pricing plan," *Electric Power Syst. Res.*, vol. 173, pp. 240–250, Aug. 2019, doi: 10.1016/j.epsr.2019.04.012.
- [140] M. S. Javadi, M. Gough, S. A. Mansouri, A. Ahmarinejad, E. Nematbakhsh, S. F. Santos, and J. P. S. Catalão, "A two-stage joint operation and planning model for sizing and siting of electrical energy storage devices considering demand response programs," *Int. J. Electr. Power Energy Syst.*, vol. 138, Jun. 2022, Art. no. 107912, doi: 10.1016/j.ijepes.2021.107912.
- [141] H. U. R. Habib, A. Waqar, M. G. Hussien, A. K. Junejo, M. Jahangiri, R. M. Imran, Y. Kim, and J. Kim, "Analysis of microgrid's operation integrated to renewable energy and electric vehicles in view of multiple demand response programs," *IEEE Access*, vol. 10, pp. 7598–7638, 2022, doi: 10.1109/ACCESS.2022.3140587.
- [142] M. F. Sulaima, N. Y. Dahlan, Z. H. Bohari, M. N. M. Nasir, R. F. Mustafa, and D. L. Nguyen, "Simultaneous load management strategy for electronic manufacturing facilities by using EPSO algorithm," *J. Mech. Eng.*, vol. 18, no. 3, pp. 193–214, 2021.
- [143] A. R. Jordehi, "A mixed binary-continuous particle swarm optimisation algorithm for unit commitment in microgrids considering uncertainties and emissions," *Int. Trans. Electr. Energy Syst.*, vol. 30, no. 11, pp. 1–20, Nov. 2020, doi: 10.1002/2050-7038.12581.
- [144] V. M. Laitsos, D. Bargiotas, A. Daskalopulu, A. I. Arvanitidis, and L. H. Tsoukalas, "An incentive-based implementation of demand side management in power systems," *Energies*, vol. 14, no. 23, p. 7994, Nov. 2021, doi: 10.3390/en14237994.
- [145] A. Sepulveda, L. Paull, W. G. Morsi, H. Li, C. P. Diduch, and L. Chang, "A novel demand side management program using water heaters and particle swarm optimization," in *Proc. IEEE Electr. Power Energy Conf.*, Aug. 2010, pp. 1–9, doi: 10.1109/EPEC.2010.5697187.
- [146] M. Pedrasa, T. D. Spooner, and I. F. MacGill, "Scheduling of demand side resources using binary particle swarm optimization," *IEEE Trans. Power Syst.*, vol. 24, no. 3, pp. 1173–1181, Aug. 2009, doi: 10.1109/TPWRS.2009.2021219.
- [147] D. Yu, X. Zhao, Y. Wang, L. Jiang, and H. Liu, "Research on energy management of a virtual power plant based on the improved cooperative particle swarm optimization algorithm," *Frontiers Energy Res.*, vol. 10, pp. 1–13, Feb. 2022, doi: 10.3389/fenrg.2022.785569.
- [148] F. R. Albogamy, S. A. Khan, G. Hafeez, and S. Murawwat, "Realtime energy management and load scheduling with renewable energy integration in smart grid," *Sustainability*, vol. 14, no. 3, pp. 1–28, 2022, doi: 10.3390/su14031792.
- [149] P.-H. Cheng, T.-H. Huang, Y.-W. Chien, C.-L. Wu, C.-S. Tai, and L.-C. Fu, "Demand-side management in residential community realizing sharing economy with bidirectional PEV while additionally considering commercial area," *Int. J. Electr. Power Energy Syst.*, vol. 116, Mar. 2020, Art. no. 105512, doi: 10.1016/j.ijepes.2019.105512.
- [150] M. Gharibi and A. Askarzadeh, "Size optimization of an off-grid hybrid system composed of photovoltaic and diesel generator subject to load variation factor," *J. Energy Storage*, vol. 25, Oct. 2019, Art. no. 100814, doi: 10.1016/j.est.2019.100814.
- [151] X. Qi, Z. Ji, H. Wu, J. Zhang, and H. Yang, "Analysis of emergency demand response levels of central air-conditioning," *J. Electr. Eng. Technol.*, vol. 15, no. 6, pp. 2479–2488, Nov. 2020, doi: 10.1007/s42835-020-00525-3.

- [152] X. Kong, S. Zhang, B. Sun, Q. Yang, S. Li, and S. Zhu, "Research on home energy management method for demand response based on chanceconstrained programming," *Energies*, vol. 13, no. 11, p. 2790, Jun. 2020, doi: 10.3390/en13112790.
- [153] A. Rezaee Jordehi, "Enhanced leader particle swarm optimisation (ELPSO): A new algorithm for optimal scheduling of home appliances in demand response programs," *Artif. Intell. Rev.*, vol. 53, no. 3, pp. 2043–2073, Mar. 2020, doi: 10.1007/s10462-019-09726-3.
- [154] H. Yang, X. Zhang, Y. Ma, and D. Zhang, "Critical peak rebate strategy and application to demand response," *Protection Control Modern Power Syst.*, vol. 6, no. 1, Dec. 2021, doi: 10.1186/s41601-021-00206-x.
- [155] H. Jabir, J. Teh, D. Ishak, and H. Abunima, "Impacts of demand-side management on electrical power systems: A review," *Energies*, vol. 11, no. 5, p. 1050, Apr. 2018, doi: 10.3390/en11051050.
- [156] T. J. Brennan and K. L. Palmer, "Energy efficiency resource standards: Economics and policy," *Utilities Policy*, vol. 25, pp. 58–68, Jun. 2013, doi: 10.1016/j.jup.2013.02.001.
- [157] (2019). Multiple Benefits of Energy Efficiency. Paris. [Online]. Available: https://www.iea.org/reports/multiple-benefits-of-energy-efficiency
- [158] F. Amirifard, S. A. Sharif, and F. Nasiri, "Application of passive measures for energy conservation in buildings—A review," Adv. Building Energy Res., vol. 13, no. 2, pp. 282–315, Jul. 2019, doi: 10.1080/17512549.2018.1488617.
- [159] M. Mukhtar, B. Ameyaw, N. Yimen, Q. Zhang, O. Bamisile, H. Adun, and M. Dagbasi, "Building retrofit and energy conservation/efficiency review: A techno-environ-economic assessment of heat pump system retrofit in housing stock," *Sustainability*, vol. 13, no. 2, pp. 1–23, 2021, doi: 10.3390/su13020983.
- [160] S. Bilgen and İ. Sarıkaya, "Energy conservation policy and environment for a clean and sustainable energy future," *Energy Sources, B, Econ., Planning, Policy*, vol. 13, no. 3, pp. 183–189, Mar. 2018, doi: 10.1080/15567249.2017.1423412.
- [161] M. A. Andor and K. M. Fels, "Behavioral economics and energy conservation—A systematic review of non-price interventions and their causal effects," *Ecological Econ.*, vol. 148, pp. 178–210, Jun. 2018, doi: 10.1016/j.ecolecon.2018.01.018.
- [162] P. Kumar, H. Caggiano, C. Cuite, C. J. Andrews, F. A. Felder, R. Shwom, K. Floress, S. Ahamed, and C. Schelly, "Behaving or not? Explaining energy conservation via identity, values, and awareness in U.S. suburban homes," *Energy Res. Social Sci.*, vol. 92, Oct. 2022, Art. no. 102805, doi: 10.1016/j.erss.2022.102805.
- [163] D. Qian, Y. Li, F. Niu, and Z. O'Neill, "Nationwide savings analysis of energy conservation measures in buildings," *Energy Convers. Manage.*, vol. 188, pp. 1–18, May 2019, doi: 10.1016/j.enconman.2019.03.035.
- [164] M. Abdul Mujeebu and F. Bano, "Integration of passive energy conservation measures in a detached residential building design in warm humid climate," *Energy*, vol. 255, Sep. 2022, Art. no. 124587, doi: 10.1016/j.energy.2022.124587.
- [165] L. Aditya, T. M. I. Mahlia, B. Rismanchi, H. M. Ng, M. H. Hasan, H. S. C. Metselaar, O. Muraza, and H. B. Aditiya, "A review on insulation materials for energy conservation in buildings," *Renew. Sustain. Energy Rev.*, vol. 73, pp. 1352–1365, Jun. 2017, doi: 10.1016/j.rser.2017. 02.034.
- [166] M. Abdul Mujeebu and F. Bano, "Energy-saving potential and costeffectiveness of active energy-efficiency measures for residential building in warm-humid climate," *Energy for Sustain. Develop.*, vol. 67, pp. 163–176, Apr. 2022, doi: 10.1016/j.esd.2022.01.011.
- [167] S. Yilmaz, A. Rinaldi, and M. K. Patel, "DSM interactions: What is the impact of appliance energy efficiency measures on the demand response (peak load management)?" *Energy Policy*, vol. 139, Apr. 2020, Art. no. 111323, doi: 10.1016/j.enpol.2020.111323.
- [168] H. G. Kim, H. J. Kim, C. H. Jeon, M. W. Chae, Y. H. Cho, and S. S. Kim, "Analysis of energy saving effect and cost efficiency of ECMs to upgrade the building energy code," *Energies*, vol. 13, no. 18, pp. 1–22, 2020, doi: 10.3390/en13184955.
- [169] B. Grillone, G. Mor, S. Danov, J. Cipriano, and A. Sumper, "A datadriven methodology for enhanced measurement and verification of energy efficiency savings in commercial buildings," *Appl. Energy*, vol. 301, Nov. 2021, Art. no. 117502, doi: 10.1016/j.apenergy.2021.117502.
- [170] Purpose of Measurement and Verification (M&V), Efficiency Valuation Organization (EVO), Washington, DC, USA, 2019. [Online]. Available: https://evo-world.org/en/m-v/purpose-of-m-v

- [171] A. B. Owolabi, B. E. K. Nsafon, J. W. Roh, D. Suh, and J.-S. Huh, "Measurement and verification analysis on the energy performance of a retrofit residential building after energy efficiency measures using RETScreen expert," *Alexandria Eng. J.*, vol. 59, no. 6, pp. 4643–4657, Dec. 2020, doi: 10.1016/j.aej.2020.08.022.
- [172] N. Oses, A. Legarretaetxebarria, M. Quartulli, I. García, and M. Serrano, "Uncertainty reduction in measuring and verification of energy savings by statistical learning in manufacturing environments," *Int. J. Interact. Design Manuf. (IJIDeM)*, vol. 10, no. 3, pp. 291–299, Aug. 2016, doi: 10.1007/s12008-016-0302-y.
- [173] C. V. Gallagher, K. Leahy, P. O'Donovan, K. Bruton, and D. T. J. O'Sullivan, "Development and application of a machine learning supported methodology for measurement and verification (M&V) 2.0," *Energy Buildings*, vol. 167, pp. 8–22, May 2018, doi: 10.1016/j.enbuild.2018.02.023.
- [174] H. Carstens, X. Xia, and S. Yadavalli, "Efficient metering and surveying sampling designs in longitudinal measurement and verification for lighting retrofit," *Energy Buildings*, vol. 154, pp. 430–447, Nov. 2017, doi: 10.1016/j.enbuild.2017.08.080.
- [175] C. V. Gallagher, K. Leahy, P. O'Donovan, K. Bruton, and D. T. J. O'Sullivan, "IntelliMaV: A cloud computing measurement and verification 2.0 application for automated, near real-time energy savings quantification and performance deviation detection," *Energy Buildings*, vol. 185, pp. 26–38, Feb. 2019, doi: 10.1016/j.enbuild.2018. 12.034.
- [176] E. Fabrizio and V. Monetti, "Methodologies and advancements in the calibration of building energy models," *Energies*, vol. 8, no. 4, pp. 2548–2574, Mar. 2015, doi: 10.3390/en8042548.
- [177] N. Li, Z. Yang, B. Becerik-Gerber, C. Tang, and N. Chen, "Why is the reliability of building simulation limited as a tool for evaluating energy conservation measures?" *Appl. Energy*, vol. 159, pp. 196–205, Dec. 2015, doi: 10.1016/j.apenergy.2015.09.001.
- [178] International Performance Measurement and Verification Protocol: Concepts and Options for Determining Energy and Water Savings, IMPVP, Efficiency Valuation Organisation, Oak Ridge, TN, USA, 2010.
- [179] ASHRAE Guideline 14–2014: Measurement of Energy and Demand Savings, ASHRAE, American Society of Heating, Refrigerating and Air-Conditioning Engineers, Atlanta, GA, USA, 2014.
- [180] D. Hou, I. G. Hassan, and L. Wang, "Review on building energy model calibration by Bayesian inference," *Renew. Sustain. Energy Rev.*, vol. 143, Jun. 2021, Art. no. 110930, doi: 10.1016/j.rser.2021. 110930.
- [181] K. Sun, T. Hong, S. C. Taylor-Lange, and M. A. Piette, "A patternbased automated approach to building energy model calibration," *Appl. Energy*, vol. 165, pp. 214–224, Mar. 2016, doi: 10.1016/j.apenergy.2015. 12.026.
- [182] S. Zuhaib, M. Hajdukiewicz, and J. Goggins, "Application of a staged automated calibration methodology to a partially-retrofitted university building energy model," *J. Building Eng.*, vol. 26, Nov. 2019, Art. no. 100866, doi: 10.1016/j.jobe.2019.100866.
- [183] G. R. Newsham, "Measurement and verification of energy conservation measures using whole-building electricity data from four identical office towers," *Appl. Energy*, vol. 255, Dec. 2019, Art. no. 113882, doi: 10.1016/j.apenergy.2019.113882.
- [184] J. Granderson and S. Fernandes, "The state of advanced measurement and verification technology and industry application," *Electr. J.*, vol. 30, no. 8, pp. 8–16, Oct. 2017, doi: 10.1016/j.tej.2017.08.005.
- [185] M. F. Zia, E. Elbouchikhi, and M. Benbouzid, "Microgrids energy management systems: A critical review on methods, solutions, and prospects," *Appl. Energy*, vol. 222, pp. 1033–1055, Jul. 2018, doi: 10.1016/j.apenergy.2018.04.103.
- [186] S. K. Rathor and D. Saxena, "Energy management system for smart grid: An overview and key issues," *Int. J. Energy Res.*, vol. 44, no. 6, pp. 4067–4109, May 2020, doi: 10.1002/er.4883.
- [187] A. Hussain, V.-H. Bui, and H.-M. Kim, "A resilient and privacypreserving energy management strategy for networked microgrids," *IEEE Trans. Smart Grid*, vol. 9, no. 3, pp. 2127–2139, May 2018, doi: 10.1109/TSG.2016.2607422.
- [188] H. A. Muqeet, R. Liaqat, M. Jamil, and A. A. Khan, "A state-of-theart review of smart energy systems and their management in a smart grid environment," *Energies*, vol. 16, no. 1, p. 472, Jan. 2023, doi: 10.3390/en16010472.

- [189] A. Al-Othman, M. Tawalbeh, R. Martis, S. Dhou, M. Orhan, M. Qasim, and A. G. Olabi, "Artificial intelligence and numerical models in hybrid renewable energy systems with fuel cells: Advances and prospects," *Energy Convers. Manage.*, vol. 253, Feb. 2022, Art. no. 115154, doi: 10.1016/j.enconman.2021.115154.
- [190] U. Previti, S. Brusca, A. Galvagno, and F. Famoso, "Influence of energy management system control strategies on the battery state of health in hybrid electric vehicles," *Sustainability*, vol. 14, no. 19, p. 12411, Sep. 2022, doi: 10.3390/su141912411.
- [191] L. Olatomiwa, S. Mekhilef, M. S. Ismail, and M. Moghavvemi, "Energy management strategies in hybrid renewable energy systems: A review," *Renew. Sustain. Energy Rev.*, vol. 62, pp. 821–835, Sep. 2016, doi: 10.1016/j.rser.2016.05.040.
- [192] Y. Zahraoui, I. Alhamrouni, S. Mekhilef, and M. R. B. Khan, "Energy management system in microgrids: A comprehensive review," *Sustain-ability*, vol. 13, no. 19, pp. 1–33, 2021, doi: 10.3390/su131910492.
- [193] N. Xu, Y. Kong, L. Chu, H. Ju, Z. Yang, Z. Xu, and Z. Xu, "Towards a smarter energy management system for hybrid vehicles: A comprehensive review of control strategies," *Appl. Sci.*, vol. 9, no. 10, p. 2026, May 2019, doi: 10.3390/app9102026.
- [194] X. Jin, K. Baker, D. Christensen, and S. Isley, "Foresee: A user-centric home energy management system for energy efficiency and demand response," *Appl. Energy*, vol. 205, pp. 1583–1595, Nov. 2017, doi: 10.1016/j.apenergy.2017.08.166.
- [195] U. Zafar, S. Bayhan, and A. Sanfilippo, "Home energy management system concepts, configurations, and technologies for the smart grid," *IEEE Access*, vol. 8, pp. 119271–119286, 2020, doi: 10.1109/ACCESS.2020.3005244.
- [196] B. Mahapatra and A. Nayyar, "Home energy management system (HEMS): Concept, architecture, infrastructure, challenges and energy management schemes," *Energy Syst.*, vol. 13, no. 3, pp. 643–669, Aug. 2022, doi: 10.1007/s12667-019-00364-w.
- [197] S. A. Mansouri, A. Rezaee Jordehi, M. Marzband, M. Tostado-Véliz, F. Jurado, and J. A. Aguado, "An IoT-enabled hierarchical decentralized framework for multi-energy microgrids market management in the presence of smart prosumers using a deep learning-based forecaster," *Appl. Energy*, vol. 333, Mar. 2023, Art. no. 120560, doi: 10.1016/j.apenergy.2022.120560.
- [198] C.-F. Chen, X. Xu, J. Adams, J. Brannon, F. Li, and A. Walzem, "When east meets west: Understanding residents' home energy management system adoption intention and willingness to pay in Japan and the United States," *Energy Res. Social Sci.*, vol. 69, Nov. 2020, Art. no. 101616, doi: 10.1016/j.erss.2020.101616.
- [199] M.-T. Ke, C.-H. Yeh, and C.-J. Su, "Cloud computing platform for real-time measurement and verification of energy performance," *Appl. Energy*, vol. 188, pp. 497–507, Feb. 2017, doi: 10.1016/j.apenergy.2016.12.034.
- [200] T. Yang, Y. Pan, J. Mao, Y. Wang, and Z. Huang, "An automated optimization method for calibrating building energy simulation models with measured data: Orientation and a case study," *Appl. Energy*, vol. 179, pp. 1220–1231, Oct. 2016, doi: 10.1016/j.apenergy.2016.07.084.
- [201] A. Cacabelos, P. Eguía, L. Febrero, and E. Granada, "Development of a new multi-stage building energy model calibration methodology and validation in a public library," *Energy Buildings*, vol. 146, pp. 182–199, Jul. 2017, doi: 10.1016/j.enbuild.2017.04.071.
- [202] X. Lü, Y. Wu, J. Lian, and Y. Zhang, "Energy management and optimization of PEMFC/battery mobile robot based on hybrid rule strategy and AMPSO," *Renew. Energy*, vol. 171, pp. 881–901, Jun. 2021, doi: 10.1016/j.renene.2021.02.135.
- [203] E. Atasoy, B. Çetin, and Ö. Bayer, "Experiment-based optimization of an energy-efficient heat pump integrated water heater for household appliances," *Energy*, vol. 245, Apr. 2022, Art. no. 123308, doi: 10.1016/j.energy.2022.123308.
- [204] S. Chaturvedi, N. Bhatt, R. Gujar, and D. Patel, "Application of PSO and GA stochastic algorithms to select optimum building envelope and air conditioner size—A case of a residential building prototype," *Mater. Today, Proc.*, vol. 57, no. 1, pp. 49–56, 2022, doi: 10.1016/j.matpr.2022.01.330.
- [205] G. Zhou, H. Moayedi, M. Bahiraei, and Z. Lyu, "Employing artificial bee colony and particle swarm techniques for optimizing a neural network in prediction of heating and cooling loads of residential buildings," *J. Cleaner Prod.*, vol. 254, May 2020, Art. no. 120082, doi: 10.1016/j.jclepro.2020.120082.

- [206] J. Zou, Q. Chang, X. Ou, J. Arinez, and G. Xiao, "Resilient adaptive control based on renewal particle swarm optimization to improve production system energy efficiency," *J. Manuf. Syst.*, vol. 50, pp. 135–145, Jan. 2019, doi: 10.1016/j.jmsy.2018.12.007.
- [207] K. Ma, M. Liu, and J. Zhang, "Online optimization method of cooling water system based on the heat transfer model for cooling tower," *Energy*, vol. 231, Sep. 2021, Art. no. 120896, doi: 10.1016/j.energy.2021.120896.
- [208] C. Andrade-Cabrera, D. Burke, W. J. N. Turner, and D. P. Finn, "Ensemble calibration of lumped parameter retrofit building models using particle swarm optimization," *Energy Buildings*, vol. 155, pp. 513–532, Nov. 2017, doi: 10.1016/j.enbuild.2017.09.035.
- [209] C.-L. Liao, S.-J. Lee, Y.-S. Chiou, C.-R. Lee, and C.-H. Lee, "Power consumption minimization by distributive particle swarm optimization for luminance control and its parallel implementations," *Exp. Syst. Appl.*, vol. 96, pp. 479–491, Apr. 2018, doi: 10.1016/j.eswa.2017.11.002.
- [210] M. H. R. Miranda, F. L. Silva, M. A. M. Lourenço, J. J. Eckert, and L. C. A. Silva, "Vehicle drivetrain and fuzzy controller optimization using a planar dynamics simulation based on a real-world driving cycle," *Energy*, vol. 257, Oct. 2022, Art. no. 124769, doi: 10.1016/j.energy.2022.124769.
- [211] K. R. Wagiman, M. N. Abdullah, M. Y. Hassan, and N. H. M. Radzi, "A new metric for optimal visual comfort and energy efficiency of building lighting system considering daylight using multi-objective particle swarm optimization," *J. Building Eng.*, vol. 43, Nov. 2021, Art. no. 102525, doi: 10.1016/j.jobe.2021.102525.
- [212] B. Su, Y. Lin, J. Wang, X. Quan, Z. Chang, and C. Rui, "Sewage treatment system for improving energy efficiency based on particle swarm optimization algorithm," *Energy Rep.*, vol. 8, pp. 8701–8708, Nov. 2022, doi: 10.1016/j.egyr.2022.06.053.
- [213] J. Chang, Z. Li, Y. Huang, X. Yu, R. Jiang, R. Huang, and X. Yu, "Multiobjective optimization of a novel combined cooling, dehumidification and power system using improved M-PSO algorithm," *Energy*, vol. 239, Jan. 2022, Art. no. 122487, doi: 10.1016/j.energy.2021.122487.
- [214] Y. Si, R. Wang, S. Zhang, W. Zhou, A. Lin, and G. Zeng, "Configuration optimization and energy management of hybrid energy system for marine using quantum computing," *Energy*, vol. 253, Aug. 2022, Art. no. 124131, doi: 10.1016/j.energy.2022.124131.
- [215] B. A. Bhayo, H. H. Al-Kayiem, S. I. U. Gilani, N. Khan, and D. Kumar, "Energy management strategy of hybrid solar-hydro system with various probabilities of power supply loss," *Sol. Energy*, vol. 233, pp. 230–245, Feb. 2022, doi: 10.1016/j.solener.2022.01.043.
- [216] S. Garip and S. Ozdemir, "Optimization of PV and battery energy storage size in grid-connected microgrid," *Appl. Sci.*, vol. 12, no. 16, p. 8247, Aug. 2022, doi: 10.3390/app12168247.
- [217] M. Lamari, Y. Amrane, M. Boudour, and B. Boussahoua, "Multiobjective economic/emission optimal energy management system for scheduling micro–grid integrated virtual power plant," *Energy Sci. Eng.*, vol. 10, no. 8, pp. 3057–3074, Aug. 2022, doi: 10.1002/ ese3.1188.
- [218] X. Lü, R. Deng, C. Chen, Y. Wu, R. Meng, and L. Long, "Performance optimization of fuel cell hybrid power robot based on power demand prediction and model evaluation," *Appl. Energy*, vol. 316, Jun. 2022, Art. no. 119087, doi: 10.1016/j.apenergy.2022.119087.
- [219] W. Li, G. Gong, Z. Ren, Q. Ouyang, P. Peng, L. Chun, and X. Fang, "A method for energy consumption optimization of air conditioning systems based on load prediction and energy flexibility," *Energy*, vol. 243, Mar. 2022, Art. no. 123111, doi: 10.1016/j.energy.2022.123111.
- [220] G. Farinis and F. D. Kanellos, "Integrated energy management system for microgrids of building prosumers," *Electric Power Syst. Res.*, vol. 198, Sep. 2021, Art. no. 107357, doi: 10.1016/j.epsr.2021.107357.
- [221] Z. Afroz, G. M. Shafiullah, T. Urmee, M. A. Shoeb, and G. Higgins, "Predictive modelling and optimization of HVAC systems using neural network and particle swarm optimization algorithm," *Building Environ.*, vol. 209, Feb. 2022, Art. no. 108681, doi: 10.1016/j.buildenv.2021.108681.
- [222] M. Beaudin and H. Zareipour, "Home energy management systems: A review of modelling and complexity," *Renew. Sustain. Energy Rev.*, vol. 45, pp. 318–335, May 2015, doi: 10.1016/j.rser.2015.01.046.
- [223] M. Rasheed, N. Javaid, A. Ahmad, Z. Khan, U. Qasim, and N. Alrajeh, "An efficient power scheduling scheme for residential load management in smart homes," *Appl. Sci.*, vol. 5, no. 4, pp. 1134–1163, Nov. 2015, doi: 10.3390/app5041134.

- [224] T. Logenthiran, D. Srinivasan, and E. Phyu, "Particle swarm optimization for demand side management in smart grid," in *Proc. IEEE Innov. Smart Grid Technol. Asia (ISGT ASIA)*, Nov. 2015, pp. 1–6, doi: 10.1109/ISGT-Asia.2015.7386973.
- [225] H. Kuo, S. Kr. Pradhan, C. Wu, P. Cheng, Y. Xie, and L. Fu, "Dynamic demand-side management with user's privacy concern in residential community," in *Proc. IEEE Int. Conf. Autom. Sci. Eng. (CASE)*, Aug. 2016, pp. 1094–1099, doi: 10.1109/COASE.2016.7743526.
- [226] M. Sim, D. Suh, and M.-O. Otto, "Multi-objective particle swarm optimization-based decision support model for integrating renewable energy systems in a Korean campus building," *Sustainability*, vol. 13, no. 15, p. 8660, Aug. 2021, doi: 10.3390/su13158660.
- [227] A. Tiwari and N. M. Pindoriya, "Automated demand response in smart distribution grid: A review on metering infrastructure, communication technology and optimization models," *Electric Power Syst. Res.*, vol. 206, May 2022, Art. no. 107835, doi: 10.1016/j.epsr.2022. 107835.
- [228] A. Ahmad, "An optimized home energy management system with integrated renewable energy and storage resources," *Energies*, vol. 10, no. 4, pp. 1–35, 2017, doi: 10.3390/en10040549.
- [229] N. Tutkun, A. Burgio, M. Jasinski, Z. Leonowicz, and E. Jasinska, "Intelligent scheduling of smart home appliances based on demand response considering the cost and peak-to-average ratio in residential homes," *Energies*, vol. 14, no. 24, p. 8510, Dec. 2021.
- [230] J. Ul Ain Binte Wasif Ali, S. A. A. Kazmi, A. Altamimi, Z. A. Khan, O. Alrumayh, and M. M. Malik, "Smart energy management in virtual power plant paradigm with a new improved multilevel optimization based approach," *IEEE Access*, vol. 10, pp. 50062–50077, 2022, doi: 10.1109/ACCESS.2022.3169707.
- [231] C.-J. Tien, C.-Y. Yang, M.-T. Tsai, and C.-Y. Chung, "The optimal operation and dispatch of commerce air-conditioning system by considering demand response strategies," *Inventions*, vol. 7, no. 3, p. 69, Aug. 2022, doi: 10.3390/inventions7030069.
- [232] M. Hughes, M. Goerigk, and T. Dokka, "Particle swarm metaheuristics for robust optimisation with implementation uncertainty," *Comput. Operations Res.*, vol. 122, Oct. 2020, Art. no. 104998, doi: 10.1016/j.cor.2020.104998.
- [233] H. Hou, Q. Wang, Z. Xiao, M. Xue, Y. Wu, X. Deng, and C. Xie, "Data-driven economic dispatch for islanded micro-grid considering uncertainty and demand response," *Int. J. Electr. Power Energy Syst.*, vol. 136, Mar. 2022, Art. no. 107623, doi: 10.1016/j.ijepes.2021. 107623.
- [234] N. Borges, J. Soares, and Z. Vale, "Multi-objective particle swarm optimization to solve energy scheduling with vehicle-to-grid in office buildings considering uncertainties," *IFAC-PapersOnLine*, vol. 50, no. 1, pp. 3356–3361, Jul. 2017, doi: 10.1016/j.ifacol.2017. 08.523.
- [235] H. Hwang Goh, S. Shi, X. Liang, D. Zhang, W. Dai, H. Liu, S. Yuong Wong, T. Agustiono Kurniawan, K. Chen Goh, and C. Leei Cham, "Optimal energy scheduling of grid-connected microgrids with demand side response considering uncertainty," *Appl. Energy*, vol. 327, Dec. 2022, Art. no. 120094, doi: 10.1016/j.apenergy.2022.120094.
- [236] M. Yang, Y. Cui, and J. Wang, "Multi-objective optimal scheduling of island microgrids considering the uncertainty of renewable energy output," *Int. J. Electr. Power Energy Syst.*, vol. 144, Jan. 2023, Art. no. 108619, doi: 10.1016/j.ijepes.2022.108619.
- [237] R. Zhang, C. Yang, and X. Li, "Optimal energy consumption optimization in a smart house by considering electric vehicles and demand response via a hybrid gravitational search and particle swarm optimization algorithm," *Energy Eng.*, vol. 119, no. 6, pp. 2489–2511, 2022, doi: 10.32604/ee.2022.021517.
- [238] G. R. Goyal and S. Vadhera, "Multi-interval programming based scheduling of appliances with user preferences and dynamic pricing in residential area," *Sustain. Energy, Grids Netw.*, vol. 27, Sep. 2021, Art. no. 100511, doi: 10.1016/j.segan.2021.100511.
- [239] A. Abbasi, K. Sultan, M. A. Aziz, A. U. Khan, H. A. Khalid, J. M. Guerrero, and B. A. Zafar, "A novel dynamic appliance clustering scheme in a community home energy management system for improved stability and resiliency of microgrids," *IEEE Access*, vol. 9, pp. 142276–142288, 2021, doi: 10.1109/ACCESS.2021.3119538.
- [240] J. Wang, Y. Li, and Y. Zhou, "Interval number optimization for household load scheduling with uncertainty," *Energy Buildings*, vol. 130, pp. 613–624, Oct. 2016, doi: 10.1016/j.enbuild.2016.08.082.

- [241] M. Partovi, S. Esmaeili, and M. Aein, "Probabilistic optimal management of active and reactive power in distribution networks using electric vehicles with harmonic compensation capability," *IET Gener, Transmiss. Distribution*, vol. 16, no. 21, pp. 4304–4320, Nov. 2022, doi: 10.1049/gtd2.12599.
- [242] H. Lotfi, "Multi-objective energy management approach in distribution grid integrated with energy storage units considering the demand response program," *Int. J. Energy Res.*, vol. 44, no. 13, pp. 10662–10681, Oct. 2020, doi: 10.1002/er.5709.
- [243] L. Li, Y. Fu, J. C. H. Fung, K. T. Tse, and A. K. H. Lau, "Development of a back-propagation neural network combined with an adaptive multiobjective particle swarm optimizer algorithm for predicting and optimizing indoor CO₂ and PM2.5 concentrations," *J. Building Eng.*, vol. 54, Aug. 2022, Art. no. 104600, doi: 10.1016/j.jobe.2022.104600.
- [244] A. Perrigot, M. Perier-Muzet, P. Ortega, and D. Stitou, "Technical economic analysis of PV-driven electricity and cold cogeneration systems using particle swarm optimization algorithm," *Energy*, vol. 211, Nov. 2020, Art. no. 119009, doi: 10.1016/j.energy.2020.119009.
- [245] Z. Cao, X. Han, W. Lyons, and F. O'Rourke, "Energy management optimisation using a combined long short-term memory recurrent neural network–particle swarm optimisation model," *J. Cleaner Prod.*, vol. 326, Dec. 2021, Art. no. 129246, doi: 10.1016/j.jclepro.2021.129246.
- [246] S. Ferahtia, H. Rezk, M. A. Abdelkareem, and A. G. Olabi, "Optimal techno-economic energy management strategy for building's microgrids based bald eagle search optimization algorithm," *Appl. Energy*, vol. 306, Jan. 2022, Art. no. 118069, doi: 10.1016/j.apenergy.2021.118069.
- [247] P. Peng, X. Li, and Z. Shen, "Energy storage capacity optimization of residential buildings considering consumer purchase intention: A mutually beneficial way," *J. Energy Storage*, vol. 51, Jul. 2022, Art. no. 104455, doi: 10.1016/j.est.2022.104455.
- [248] W. Zhao, Z. Luan, and C. Wang, "Parametric optimization of novel electric-hydraulic hybrid steering system based on a shuffled particle swarm optimization algorithm," *J. Cleaner Prod.*, vol. 186, pp. 865–876, Jun. 2018, doi: 10.1016/j.jclepro.2018.03.180.
- [249] M. Kheshti, L. Ding, S. Ma, and B. Zhao, "Double weighted particle swarm optimization to non-convex wind penetrated emission/economic dispatch and multiple fuel option systems," *Renew. Energy*, vol. 125, pp. 1021–1037, Sep. 2018, doi: 10.1016/j.renene.2018.03.024.
- [250] C. J. Meinrenken and A. Mehmani, "Concurrent optimization of thermal and electric storage in commercial buildings to reduce operating cost and demand peaks under time-of-use tariffs," *Appl. Energy*, vol. 254, Nov. 2019, Art. no. 113630, doi: 10.1016/j.apenergy.2019. 113630.
- [251] Z. Wang, Y. Gu, H. Liu, and C. Li, "Optimizing thermal–electric load distribution of large-scale combined heat and power plants based on characteristic day," *Energy Convers. Manage.*, vol. 248, Nov. 2021, Art. no. 114792, doi: 10.1016/j.enconman.2021.114792.
- [252] C. Mokhtara, B. Negrou, A. Bouferrouk, Y. Yao, N. Settou, and M. Ramadan, "Integrated supply-demand energy management for optimal design of off-grid hybrid renewable energy systems for residential electrification in arid climates," *Energy Convers. Manage.*, vol. 221, Oct. 2020, Art. no. 113192, doi: 10.1016/j.enconman.2020. 113192.
- [253] X. Fan, Z. Gao, W. Ji, L. Guo, W. Lin, and J. Wang, "Thermodynamic optimization with multi objectives and parameters for liquid air energy storage system based on the particle swarm optimization (PSO)," *J. Energy Storage*, vol. 41, Sep. 2021, Art. no. 102878, doi: 10.1016/j.est.2021.102878.
- [254] M. S. Javed, T. Ma, J. Jurasz, F. A. Canales, S. Lin, S. Ahmed, and Y. Zhang, "Economic analysis and optimization of a renewable energy based power supply system with different energy storages for a remote island," *Renew. Energy*, vol. 164, pp. 1376–1394, Feb. 2021, doi: 10.1016/j.renene.2020.10.063.
- [255] H. Bahlawan, M. Morini, M. Pinelli, P. R. Spina, and M. Venturini, "Simultaneous optimization of the design and operation of multigeneration energy systems based on life cycle energy and economic assessment," *Energy Convers. Manage.*, vol. 249, Dec. 2021, Art. no. 114883, doi: 10.1016/j.enconman.2021.114883.
- [256] B. Sun, "A multi-objective optimization model for fast electric vehicle charging stations with wind, PV power and energy storage," *J. Cleaner Prod.*, vol. 288, Mar. 2021, Art. no. 125564, doi: 10.1016/j.jclepro.2020.125564.

- [257] P. A. Gbadega and Y. Sun, "A hybrid constrained particle swarm optimization-model predictive control (CPSO-MPC) algorithm for storage energy management optimization problem in micro-grid," *Energy Rep.*, vol. 8, pp. 692–708, Nov. 2022, doi: 10.1016/j.egyr.2022.10.035.
- [258] K. K. P. Kumar, N. Soren, A. Latif, D. C. Das, S. M. S. Hussain, A. Al-Durra, and T. S. Ustun, "Day-ahead DSM-integrated hybrid-power-management-incorporated CEED of solar thermal/wind/wave/BESS system using HFPSO," *Sustainability*, vol. 14, no. 3, p. 1169, Jan. 2022, doi: 10.3390/su14031169.
- [259] A. Nawaz, G. Hafeez, I. Khan, K. U. Jan, H. Li, S. A. Khan, and Z. Wadud, "An intelligent integrated approach for efficient demand side management with forecaster and advanced metering infrastructure frameworks in smart grid," *IEEE Access*, vol. 8, pp. 132551–132581, 2020, doi: 10.1109/ACCESS.2020.3007095.
- [260] S. Dayalan, S. S. Gul, R. Rathinam, G. F. Savari, S. H. E. A. Aleem, M. A. Mohamed, and Z. M. Ali, "Multi-stage incentive-based demand response using a novel Stackelberg-particle swarm optimization," *Sustainability*, vol. 14, no. 17, p. 10985, Sep. 2022, doi: 10.3390/su141710985.
- [261] S. Pardhi, M. El Baghdadi, O. Hulsebos, and O. Hegazy, "Optimal powertrain sizing of series hybrid coach running on diesel and HVO for lifetime carbon footprint and total cost minimisation," *Energies*, vol. 15, no. 19, p. 6974, Sep. 2022, doi: 10.3390/en15196974.



FARAH ANISHAH ZAINI received the bachelor's degree in electrical engineering from Universiti Teknikal Malaysia Melaka (UTeM), in 2022, where she is currently pursuing the master's degree in electrical engineering. She is also a Graduate Research Assistant with the Energy and Power System Research Group, UTeM. Her research interests include demand response, demand side management, load forecasting, and artificial intelligence.



MOHAMAD FANI SULAIMA (Member, IEEE) received the bachelor's degree from Tokai University, Japan, in 2010, the master's degree from the University of Malaya, and the Ph.D. degree in electrical engineering with a specialization in energy demand side management from Universiti Teknologi Mara (UiTM), Malaysia, in 2020. He is a Senior Lecturer with the Faculty of Electrical Engineering, Universiti Teknikal Malaysia Melaka (UTeM). Upon joining UTeM, he was a Coordina-

tor and the Head of the Energy Management Division, Centre for Sustainability and Environment, before appointed as the first Internal University Energy Manager, in 2015. He is also the Head of the Research Laboratory for Energy and Power Systems. His research interests include power systems, demandside management, demand response, energy efficiency, measurement and verification, and artificial intelligence. As a result of his research interest, he has published more than 100 articles, including high-impact journals and academic papers.



INTAN AZMIRA WAN ABDUL RAZAK received the B.Sc. degree in electrical engineering from Universiti Teknikal Malaysia Melaka (UTeM), Malaysia, in 2006, the M.Eng. degree in electrical power from Universiti Teknologi Malaysia (UTM), in 2008, and the Ph.D. degree from The National Energy University (UNITEN), Malaysia, in 2017. Currently, she is a Senior Lecturer with the Faculty of Electrical Engineering, UTeM. Her main research interests include load and price fore-

casting, artificial intelligence, and optimization algorithm.



HAZLIE MOKHLIS (Senior Member, IEEE) received the B.Eng. and M.Eng.Sc. degrees in electrical engineering from the University of Malaya (UM), Malaysia, in 1999 and 2002, respectively, and the Ph.D. degree from The University of Manchester, U.K., in 2009. He is currently a Professor with the Department of Electrical Engineering, UM. He is also the Head of the UM Power and Energy System (UMPES) Research. His research interests include fault loca-

tion, distribution automation, power system protection, and renewable energy. He is also a Chartered Engineer in the U.K. and a Professional Engineer in Malaysia.

...



NUR IZYAN ZULKAFLI received the B.Eng. and M.Sc. degrees in chemical engineering from Universiti Teknologi PETRONAS, Malaysia, in 2009 and 2011, respectively, and the Ph.D. degree in energy and power from Cranfield University, U.K., in 2019. Currently, she is a Senior Lecturer with the Faculty of Mechanical Engineering, Universiti Teknikal Malaysia Melaka (UTeM) and also a Coordinator of the master project taught course. In research, she has actively involved in optimiza-

tion modeling activities for research projects in the area of production and energy system for buildings and process industries. Her research interests include efficient energy and power, operational and maintenance planning, supply chain, and demand response program. Throughout her research activities, she has managed to produce high-quality research articles mainly in high-indexed journals.